Initial Environmental Examination

July 2016

LOAN 3173-VIE: INTEGRATED RURAL DEVELOPMENT SECTOR PROJECT IN THE CENTRAL PROVINCES (Additional Financing)

SUBPROJECT: UPGRADING IRRIGATION/ DRAINAGE AND RURAL ROAD SYSTEM IN FLOODING-PRONE AREA OF LOC HA DISTRICT

CURRENCY EQUIVALENTS

(As of 20 May 2016)

Currency unit _ Vietnamese Dong (VND)

VND 1.00 = \$0.0000449 \$1.00 VND 22,275 =

ABBREVIATIONS

ADB Asian Development Bank

AΡ Affected persons

CPC Communal People's committee CPMU Central Project Management Unit

DARD Department of Agriculture and Rural Development DONRE Department of Natural Resources and Environment

DPC District People's Committee

EIAR **Environmental Impact Assessment Report EMDF** Ethnic Minority Development Framework

EMP Environmental Management Plan Environmental Protection Plan

EPP

DARD Department of Agriculture and Rural Development

FPD Forest Protection Department

IEE Initial Environmental Examination

IPM Integrated Pest Management

IRDSPCP Integrated Rural Development Sector Project in Central Provinces

LIC Loan Implementation Consultant

Ministry of Natural Resources and Environment MONRE

PC People's Committee

PPC Provincial Peoples Committee

PPMU Provincial Project Management Unit

RF Resettlement Framework

SIR Subproject Investment Report

TPC Town People's Committee

The IRDSPCP - Additional Financing The Projects

UXO **Unexploded Ordnance**

WEIGHTS AND MEASURES

km – kilometer

kg – kilogram

ha – hectare

m – meter

NOTE

In this report, "\$" refers to US dollars.

This initial environmental examination is a document of the borrower. The views expressed herein do not necessarily represent those of ADB's Board of Directors, Management, or staff, and may be preliminary in nature. Your attention is directed to the "terms of use" section of this website.

In preparing any country program or strategy, financing any project, or by making any designation of or reference to a particular territory or geographic area in this document, the Asian Development Bank does not intend to make any judgments as to the legal or other status of any territory or area.

TABLE OF CONTENTS

LIST OF TABLES		II
LIST OF PHOTOS		II
1. INTRODUCTION		3
2. PROJECT DESCRIPTION	l	5
3. DESCRIPTION OF EXIST	ΓING ENVIRONMENT	12
4. ENVIRONMENTAL IMPAG	CT SCREENING	15
4.1 Environmental Mitigation	n Plan	30
4.2.1 Environmental effects	ng Plan monitoring iance Monitoring	37
4.3 EMP Implementation Arr	rangements	45
4.4 Monitoring and Reportin	ng System	48
4.5 EMP Budget		49
5. PUBLIC CONSULTATION	N AND DISCLOSURE ACTIVITIES	49
5.1 Description of Activities	to Date	49
5.2 Outcomes of Public Con	nsultation to Date	50
<u>-</u>	Itation for the preparation of Environmental Impact	51
5.2 Future Public Consultati	ion Activities	51
6. CONCLUSION AND RECO	OMMENDATIONS	52
7 ANNEYEO		

LIST OF TABLES

Table 1 - General information of subproject	5
Table 2. Environmental baseline	12
Table 3. Environmental impact screening	15
Table 4 - Environmental mitigation plan	30
Table 5. Environmental effects monitoring plan	37
Table 6. Environmental Compliance Monitoring	40
Table 7 - EMP Implementation	45
Table 8 -Monitoring and Reporting System	48
Table 9 - EMP Budget	49
Table 10 - Public consultation and public disclosure activities	49
Table 11 - Results of public consultation	50
Table 12 - Proposed community consultation activities	51
LIST OF PHOTOS	
Photo 1: The existing spillway of	55
Photo 2: Overview of Lower Khe Hao Reservoir	55
Photo 3: Water quality monitoring point at Khe Hao reservoir	55
Photo 4: Air quality monitoring point at the beginning of canal no.3	55
Photo 5: Air quality monitoring point in front of Lu temple, Tan Loc commune	56
Photo 6: Air quality monitoring point in front of Hong Tan secondary school, commune	
Photo 7: A section in Loi Ma – Cau Ngao drainage canal	56
Photo 8: Water quality monitoring point in Tan Thuong hamlet, Tan Loc commune	56
Photo 9: Current condition of the Khe Hao Management Road	56
Photo 10: Water quality monitoring point in Kim Tan hamlet, Tan Loc commune	56
Photo 11: Public consultation in Tan Loc commune	73
Photo 12: Public consultation in Tan Loc commune	73

1. INTRODUCTION

- Loan 2357(SF) for the Integrated Rural Development Sector Project in the Central Provinces (IRDSPCP) was approved by ADB on 15 October 2007. The total cost of the Project was estimated at \$168.2 million and is jointly financed by ADB and Agence Française de Développement. The IRDSPCP focuses on upgrading and rehabilitating rural infrastructure (rural roads and irrigation systems, flood control, markets and other key infrastructure). To date, 129 subprojects have either been completed or are nearing completion. Review missions had determined that the quality of construction of subprojects was good. The executing agency (EA) has developed the expertise needed to effectively implement the project and significant benefits are already accruing.
- 3. At the request of the Government, the potential for additional financing was investigated during the Mid-Term Review in 2011 and two review missions in 2012. About 39 new subprojects were found eligible for consideration in the additional financing. The amount of \$70 million has been recommended and included in the country program for ADB Board consideration in 2013. The IRDSPCP - Additional Financing (The Project) aims to rehabilitate and upgrade deteriorated critical productive rural infrastructure in support of the Government of Viet Nam's new National Target Program for Rural Development.
- In consultation with the relevant provincial government and field investigation by the CPMU, a total of 23-24 eligible subprojects were initially identified based on 7 screening criteria which are focused on social economic development, safeguards, integrated development model, feasibility and sustainability. The types of subprojects are as follows:
 - (i) Small & medium-sized dam and reservoir improvements e.g., spillways, head-works, reservoir walls, and leakage control;
 - (ii) Rehabilitation of primary and secondary irrigation canals and riverbank stabilization. Wherever possible key strategic investments such as the lining of critical lengths of canal or the reinforcing of existing water control structures will be chosen; and
 - (iii) Rehabilitation of commune to district, and inter-commune roads to improve linkages between higher level alignments (provincial and national routes) and lower level commune to village and inter-village roads. In addressing key issues of sustainability, designs will take into account the increased intensity and frequency of climatic hazards anticipated to result from global climate change, the local geology and terrain, potential change in utilization patterns (type and volume of traffic), and the longer-term availability of recurrent expenditure for operations and maintenance (O&M).
- As part of the Project, "UPGRADING IRRIGATION/ DRAINAGE AND RURAL ROAD SYSTEM IN FLOODING-PRONE AREA OF LOC HA DISTRICT" subproject will be implemented at Tan Loc, Binh Loc, Thinh Loc, Hong Loc and An Loc communes - Loc Ha district, Ha Tinh province.
- This Initial Environmental Examination document has been prepared to meet the environmental safeguards requirements of the ADB¹ and GOV². The IEE contains the following information:
 - Section 1 Introduction; (i)
 - Section 2 contains a description of the subproject; (ii)
 - Section 3 contains a description of environmental conditions in the vicinity of the (iii) subproject;
 - Section 4 contains a describes potential environmental impacts of the subproject; (iv)

¹ ADB Safeguard Policy Statement (2009)

² Law on Environment Protection 2014; Decree 18/2015/ND-CP and Circular 27/2015/TT-BTNMT

- (v) Section 5 contains the environmental management plan including mitigation measures, monitoring system and cost estimation for Environmental Monitoring System (EMS) implementation;
- (vi) Section 6 contains activities description on community consultation and subproject disclosure;
- (vii) Section 7 contains conclusion and recommendation including summarization of main impacts and typical mitigation measures in the subproject's implementation.

2. PROJECT DESCRIPTION

Table 1 - General information of subproject

DATA ITEM	SUBPROJECT DATA
GENERAL INFORMATION	
Subproject Name	Upgrade irrigation/ drainage and rural road system in flooding-prone area of Loc Ha district
Subproject Type	Irrigation/ Drainage/ Rural transportation
Executing Agency	People's Committee of Ha Tinh Province
Sub-project owner	Department of Agriculture and Rural Development, Ha Tinh Province
Sub-project Management Unit	PPMU of IRDSPCP, Ha Tinh Province
Address of PPMU's office	South of Cay bridge, Thach Trung commune, Ha Tinh City, Ha Tinh province
Name and Title of Head of	Mr Nguyen Van Duong
Project owner	Title: Director
Telephone, fax and email details of Project owner	Tel: 0393690336; FAX: 0393858921
Name of Environmental Officer of PPMU	Nguyen Hong Son
Telephone, fax and email details of PPMU Environmental Officer	0943678802, sonkhoa87@gmail.com
SUBPROJECT DESCRIPTION	
New project or rehabilitation project	Rehabilitation project
Surface water or groundwater source	Surface water source
Identification of water source	Upper Khe Hao; Lower Khe Hao; and Dong Ho reservoirs
Identification of drainage basin area	Upper Khe Hao reservoir: the capacity of Upper Khe Hao reservoir is 3.8 million m³, which supplies water for 29,338 people in Thach Bang, Thach Kim, Thinh Loc communes with capacity of 4000 m³/day and provide irrigation water through Lower Khe Hao reservoir.
	Lower Khe Hao reservoir: the capacity is 1.293 million m ³ . It receives water from Upper Khe Hao reservoir and provides irrigation water for 228 ha (at design capacity) cultivation area of Tan Loc and Thinh Loc communes through 15.5 km irrigation canal.
	Dong Ho reservoir: the capacity is 1.271 million m ³ . It provides irrigation water for 190 ha (at design capacity) cultivation area of Hong Loc and Binh Loc communes through 22.5 km irrigation canal.
	Hong Tan drainage canal (7 km in length) locate in Tan Loc commune, is the main drainage canal for Tan Loc, Hong Loc, Binh Loc and An Loc communes, Loc Ha district.
Type of Subproject	Repairing, upgrading reservoir and irrigation canals and rural road infrastructures
Purpose and needs of the subproject	The current/ design capacities of Khe Hao and Dong Ho irrigation systems are 64/ 228 ha and 92/ 190 ha respectively. The subproject will upgrade Lower Khe Hao reservoir and 9,466 m irrigation canal to

DATA ITEM	SUBPROJECT DATA				
	ensure the irrigation of the reservoirs, increat people.	apacity, as in	the design, of		
	Loi Ma – Thien Thinh and Tan Loc – Binh Loc are earth roads we road surface width is varied from 3-6m. It makes dust in the dry sand slippery, muddy condition when it is raining, make many different for the movement of local people especially people in Tacommune must travel 5.3 km (instead of 2.5 km due to baccondition) to reach the district market in Binh Loc commune.				
	Upgrade the drainag chance of access to r disasters likes drough development of 5 com	ural infrastruct t and flood, su	ure, minimize pport sustaina	impacts from natural	
Components of the subproject	Sub-project will repair irrigation and improving Those activities experimental improve socio-economics.	ng rural transpected to incre	oortation by up ease agricultu	ograding rural roads. Iral productivity and	
	The details of compo	onents include:			
				of Lower Khe Hao nage canals system;	
	Loc inter-commu	ne road to Vil	lage No.1 – A	ed i) Tan Loc – Binh An Loc commune, ii) ent road of Khe Hao	
a) Repairing and upgrading irrigation and drainage system of Lower Khe Hao reservoir	age and the auxiliary weir with L = 128 m. Level up the weirs from 7m				
	- Number of intakes and outlets need to constructed: 02 will be constructed in Lower Khe Hao reservoir. The first culvert will be constructed at the main weir area, 7.7 m to the left of the old culvert. The second culvert will be constructed on the right side of spillway, under the auxiliary weir. Both water intake culverts will be constructed with dimension 0.8x1.0 m width and height. Culvert wall is steel-enforced concrete R _{b2} 00, 30 cm in thickness. The slope ratio at the bottom is i=0.008;				
	- Irrigation canals: The existing canals are mainly earth canals. Subproject will concrete the irrigation canal system of Lower Khe Hao and Dong Ho Reservoir. Concrete 4,893 m irrigation canal of Khe Hao Irrigation System and 4,553 m irrigation canal of Dong Ho Irrigation System with R _b 200 concrete, stone (1x2) cm. Improve the canal bank with concrete slab R _b 200, 60x60 cm and 8 cm in thickness. The canal bed will be cover with geo-technical canvas.				
	Canal	Length (m)	Width (m)	Depth (m)	
	Canals of Khe Hao I		•		
	Canal after inlet	161.00	1.00	1.10	
	Canal no. 1	2,380.0	0.60	0.70	
	Canal no. 2	604.00	0.60	0.60	
	Canal no. 3	664.00	0.40	0.50	

DATA ITEM	SUBPROJECT DATA					
	Canal no. 4 374.00 0.40 0.40					
	Canal no. 5	534.00	0.40	0.50		
	Canal no. 6	780.00	0.50	0.60		
	Canals of Dong Ho Irrigation System					
	Canal no. 1	1419.00	0.4	0.4		
	Canal no. 2	449.10	0.4	0.4		
	Canal no. 3	615.30	0.4	0.4		
	canal bank erosion. canal, the start poi the end point links Nghen River main drainage canal. Th (near Lu Temple),	have seriously . Upgrade (i) 2 int connects w s with Dong D canal system e start point c the end point l nen River mair	degraded wit, 484 m Loi Ma, 484 m Loi Ma ith road crossion main drain; (ii) 890 m Tonnects with rong canal system	rstem is mainly earth h sedimentation and — Cau Coc drainage ing drainage culvert, hage canal - part of an Loc — Binh Loc road crossing culvert g Tan main drainage h. The canals will be		
	Canal	Length (m)	Width (m)	Depth (m)		
	Dinh Lu – Cau Ngao	890	1.6	1.4		
		2484.50				
	Loi Ma - Cau Coc	2404.30	1.2 - 1.4	1.2 - 1.4		
	Both canals system will be upgraded with trapezoid shape. Canal bank will be upgraded with mortar mixed & masonry $R_{\nu}100$, 25 cm in thick and the dimension (in average) is 1.4 x 1.5 m width and height					
b) Repairing and upgrading	Rural road: Upgrading	the following i	tems:			
rural roads	(i) Tan Loc – Binh Loc inter-commune road to Village No.1 – An Loc commune with the total length of 3,588.3 m to Rural Road Grade A (TCVN10380: 2014) with road surface of 5m, road foundation with of 6.5m, road side width of 2x0.75. Road surface material is cement concrete – 20 cm in thickness. The start point connects with bitumen road of Tan Loc commune (Dinh Lu) and the end point connects with the main path of No.1 hamlet, An Loc commune;					
	(ii) Loi Ma – Thien Thinh road with the total length of 2,324.7 m. The start point is the junction with concrete road to Lower Khe Hao reservoir, the end point connects with Thien Thinh road (Cau Coc) in Tan Loc commune;					
	(iii) Management road of Khe Hao reservoir with the total length of 1,490.2 m, which will be upgraded to rural road Grade B (TCVN10380: 2014) with surface of 3.5 m, foundation with of 5m, side width of 2x0.75. Road surface material is cemented concrete – 18 cm in thickness.					
	Both (ii) and (iii) will be 2014) with road surfact width of 2x0.75. Road thickness.	ce of 3.5m, roa	nd foundation v	with of 5m, road side		
Supplementary items	Construct new free flow water intake culverts in					
	Spill way: Spill thresho	old elevation is	5.75 m, desigr	n height is 0.443 m		

DATA ITEM	SUBPROJECT DATA		
	and length is 40 m		
CONSTRUCTION ACTIVITIES	3		
Commencement date (month/year)	Quarter IV 2016 (as expected)		
Completion date (month/year)	The 31 st of October 2018 (24 months - as expected)		
Number of workers	About 300 workers		
Necessary camps (Yes/No)	Yes, 01 main camp and 05 temporary camps along the proposed canals and roads		
Construction in rainy season (Yes/No)	Yes (the construction will also be implemented in the dry days in the rainy season to meet the deadline)		
Concrete mixing plant	No concrete mixing plant, but 250-500-litre mobile concrete mixing equipment could be used. These mobile plants need to have legal operation licenses and the contractor will discuss with local authorities before mobilize the plants to the construction site.		
Location and area of borrow area or description of material source	Filled soil will be taken from Tung Son borrow pit, Tung Loc commune and Hong Loc borrow pit, Hong Loc commune, those are located at a distance of 7 km from the subproject site. These borrow pits have the operation licensed from Ha Tinh DONRE and have been used as soil sources for construction material in Loc Ha district. The contractor will work with Tung Loc and Hong Loc CPCs before the construction commencement.		
	Stone will be purchased at construction material shop in Can Loc town (sources from quarry of Hong Linh town). It is far from the subproject area about 8 Km;		
	Sand will also be purchased at construction material shop in Can Loc town (sources from mine in Vuong Loc commune, Can Loc district).		
	Other material such as steel, cement will be purchased at Construction Material Shops in Loc Ha town located 7 km from the subproject site.		
Method on management and balance of excavated soil/surplus soil	The quantity of excavated soil along the route of the canal will not great which is used for filling back the banks, surplus soil is used for levelling the canal safety corridor;		
	Cutting soil: 23,596 m³; Filling soil: 30,653m³; Spoiled soil: 18,741m³; About 20,057 m³ of cutting soil will be reused to fill up the canal's banks and 10,267 m³ of filling soil will be taken from Tung Son and Hong Loc borrow pits.		
	Excavated soil that could not be reused will be transfer to Loc Ha landfill – about 5 km from the subproject site. This is the landfill of Loc Ha district with the total area of 5ha. The landfill start operates in January 2015 with the permission from Ha Tinh PPC. The contractor will work with Loc Ha DPC before the construction commencement for a dumping agreement.		

-

 $^{^3}$ Source: Basic Design Explanation of Lining the irrigation and drainage canal systems and road of the flooded area in Loc Ha district subproject, Ha Tinh Province, 2014

DATA ITEM	SUBPROJECT DATA					
Type and approximate quantity of raw construction materials	bel	mand of raw constructio ow table, since the sour nted above				
	Items			Unit	Quantity	
		Filling soil		m ³	30,653	
		Dry masonry		m ³	1257	
		Construction stone		m ³	8020	
		Macadam		m ³	8522	
		Concrete		m ³	5748	
		Concrete with steel			4277	
		Steel		Ton	128	
(calculated m³/month) - Soil, sand, debris, etc Domestic waste Number and conditions	kg/	mestic waste: 0.5kg/persomonth e construction equipment/	machines	could be	listed ad 03 bu	Ildozers
construction vehicles and equipment OPERATION AND MAINTENA	mix are	0Cv); 06 excavators; 03 kw); more than 6 tip trucking machines (250-500l); in good conditions and had a ACTIVITIES ⁴	ks (self-di 06 water	ump – 14ı trucks. All	m³), 26 mobile o vehicles and m	concrete achines
Water discharge at the		Canal	116	nit	Discharge (C))
beginning section of the canal:		Canal after inlet		/s	0.30	×)
(m^3/s)						
		Canal no. 1		/s	0.09	
		Canal no. 2	m	/s	0.04	
		Canal no. 3	m	/s	0.04	
		Canal no. 4	m	/s	0.03	
		Canal no. 5	m	/s	0.03	
		Canal no. 6	m	/s	0.07	
Irrigation area (ha)	Current capacity/ expected capacity (cultivation area in hectare): Khe Hao irrigational system is 64/228 – 28% and Dong Ho irrigation system is 92/190 – 49%					
Operation, management and maintenance Unit	Northern Ha Tinh Irrigation One Member Limited Company is responsible for operation, management and maintenance the reservoirs and irrigation system after completion of the work					
Water treatment process	No. Water in Lower Khe Hao reservoir has been used only for irrigation purpose without sign of water pollution. There has not been any water treatment system in the subproject area.					
Descriptions of periodical maintenance activities	ma	Annual maintenance: Annual maintenance activities include greasing machines, replacing broken minor parts of machines such as penstocks; dredging sludge of canals, culverts, spillways; repairing				

⁴ Source: SIP of lining the irrigation and drainage canal systems and road of the flooded area in Loc Ha district, 2014

DATA ITEM	SUBPROJECT	Γ DATA	
	banks of canals; removing rubbish in the canals, culverts, troughs, spillways etc. repairing minor breaks of canal slopes, culverts, spillways and treating extrusion joints, penetration through animal holes and nests of white ants. Annual average cost for Operation and Maintenance Activities estimated 1,513,842,493 VND		
	Periodic maintenance (every 7 years): Periodic maintenance is conducted every seven years for canals. Estimated cost for periodic maintenance is 30% of the capital cost including repairing significantly damaged works. Detailed activities will be based on the Economic-Technical Report according to the government's regulations on investment management.		
RESETTLEMENT AND LAND	ACQUISITION ⁵		
Affected households	0 households		
Number of severely affected APs	There is no production land loss of >10	% for any household.	
Number of APs that must relocate	No house/shop shall be relocated or rebuilt at new place.		
Total land area to be acquired (ha)	Temporary= 0 m ²	Permanent = 0 m ²	
Agricultural land area to be acquired (ha)	Temporary= 0 m ²	Permanent = 0 m ²	
Forestry land area to be acquired (ha)	Temporary= 0 m ²	Permanent = 0 m ²	
Aqua cultural land to be acquired (ha)	Temporary= 0 m ²	Permanent = 0 m ²	
Residential land to be acquired (ha)	Temporary= 0 m ²	Permanent = 0 m ²	
Garden land to be acquired (ha)	Temporary= 0 m ²	Permanent = 0 m ²	
Other land to be acquired (ha)	Temporary= 0 m ²	Permanent = 0 m ²	
Affected assets	None	None	
SUBPROJECT COST	3	-	
Total subproject cost (VND and \$USD)	79,623,946,493 VND/ 3,791,616.43 US	D (1 USD = 21000 VND)	

-

 $^{^{\}rm 5}$ This data is obtained from Resettlement Plan of the subproject

12 bach dan Legend: N2 Proposed drainage canal K3-Water quality monitoring N1 xã Thịnh Lộc N3 point at Khe Hao reservoir Water quality monitoring N2 point at canal connected to the inlet Air and water quality 1,7. Material K3monitoring point at the transportation beginning of canal no.3 road no. 22-12 xóm 2 and water quality Main canal and monitoring point in front of material temple, Tan Loc transportation commune (Proposed commune's drainage canal) road xã Tân Lôc Water and water quality K5monitoring point in Kim Tan **Proposed** hamlet, Tan Loc commune drainage canal (Proposed drainage canal) xã An Lộc - . . xóm 4 xã Hồng Lộc Tân Thượng xã Bình Lộc Tân Trung

Figure 1: Map of Subproject and Surrounding Area

3. DESCRIPTION OF EXISTING ENVIRONMENT

Table 2 - Environmental baseline

DATA ITEM	SUBPROJECT DATA
PROJECT LOCATION	JOBI ROJECT DATA
Commune(s):	Tan Loc, Thinh Loc, An Loc, Hong Loc and Binh Loc
District:	Loc Ha
Province:	Ha Tinh
Geographic location:	From 18°28'31" N ÷ 18°30'29" N.
Coograpino iocation.	From 105°50'38" E ÷ 106°52'03" E.
PHYSICAL ENVIRONMENT CO	
Air quality, noise and vibration	The ambient air quality in the subproject area is still in good quality and showed no trace of pollution. According to the ambient air quality monitoring for environment impact assessment of the subproject (which need to follow government requirement), CO, NO ₂ , SO ₂ and dust concentration are all within the allowed level of Vietnamese standard QCVN 05:2013/BTNMT on ambient air quality. Noise level varied from 54 – 58 dBA, lower than the permitted level of 70 dBA of QCVN 26: 2010/BTNMT on noise level ⁶ . The monitoring locations are near Lu temple in Tan Loc commune; Hong Tan secondary school; canal No.3 in Kim Tan village, Tan Loc commune and Kim Tan inter-village road in Tan Loc commune.
Climate and natural disasters	The subproject area has a monsoon tropical climate, divided into two distinct seasons. The rainy season lasts from April to October; average temperature fluctuates above 25°C accompanied by heavy rain. The dry season is from November to March, average temperature fluctuates below 23°C.
	The average annual precipitation is 2661 mm. September is usually the month with the highest precipitation of the year (reached up 1450 mm).
	The meteorology-hydrology data of Ha Tinh Centre indicate there are 4-5 storms annually in the subproject area with heavy rainfall. However, there are not so many storms happen in the subproject in the recent years. The nearest storm with large impact on the district was happened in September 2013 with 216 houses with roof off; flooded 203 houses; 14.25 km canal has been broken; the inundation areas were 250 ha of rice field, 110 ha of sweet potato, 120 ha of vegetables and 707 ha salt production.
Topography and soils	Subproject is locate in the narrow coastal plain so the area is generally flat, with elevations ranging from 10.0m to 50.0m and the lowest point is 2.5m. The soils are mainly silt and sand.
Waterbodies	The main water bodies in the subproject area are upper and lower Khe Hao reservoirs, Dong Ho reservoir and Nghen river. There are also water rice field distribute separated in the subproject area.
	Water from small streams in Hong Linh mountain area discharge into upper Khe Hao reservoir, which are using for domestic water supply purpose and providing water for lower Khe Hao reservoir for irrigation purpose. Dong Ho reservoir also receives water from streams in Hong Linh mountain area before irrigating through the canal system.
	There are some drainage systems in the subproject area. Water flow

⁶ The monitoring has been implemented in 25 May 2016 by Irrigation Science and Technical Implementation Institute

DATA ITEM	SUBPROJECT DATA			
	downstream then discharge into Nghen River (part of La River) before flow to the sea. Water of the river and canals is mainly used for agriculture purpose.			
Underground water	According to the underground water monitoring result for environment impact assessment of the subproject, some main parameters like TSS, CI, COD, As, Fe, are all within the allowed level of Vietnamese Standard QCVN 09:2008/BTNMT on underground water quality. However, E.coli and Coliform parameters are all higher than the allowed level with E.coli from 3 – 6 in compare with QCVN 09:2008/BTNMT and Coliform is 8 -13 in compare with QCVN 09:2008/BTNMT showed that underground water in the subproject area has been polluted with microbiology. The sampling locations are drill wells in living area of Hong Tan secondary school; Lu temple, Tan Loc commune; household near Kim Tan inter-village road in Tan Loc commune; near Canal No.3 in Tan Loc commune. ⁷			
Water quality	According to the surface water monitoring result for environment impact assessment of the subproject, some main parameters like DO, Cl, BOD $_5$, As, Fe, are all within the allowed level of Vietnamese Standard QCVN 08:2008/BTNMT on surface water quality. However, TSS parameters are higher than the allowed level at all 5 monitoring locations (varied from 55.5 – 62.6 mg/l in compare with 50 under QCVN 08:2008/BTNMT). The sampling locations are Lower Khe Hao reservoir in Tan Loc commune (2 locations); irrigation canal in Tan Thuong, Tan Loc; and drainage canals in Tan Thuong, Tan Loc commune (footnote 7).			
Flooding and drought	This area always has been in state of inundation, 2 times per year (mainly in rainy season from August to Oct). The nearest big flood was happened in October 2010 with 3230 households in 9 over 13 communes of Loc Ha district have suffered from inundation.			
Terrestrial flora and fauna	Terrestrial flora could be mainly rice field, corn, bean and vegetables gardens in residential areas. Terrestrial fauna are cow, pig, chicken, ducks, etc.			
	None of terrestrial flora and fauna in subproject area is listed in Vietnam's Red Data Book.			
Protected areas	In subproject area, there is no historical or historical vestiges or protected area;			
Environmental sensitive points	- Residential area are Kim Tan and Tan Thuong hamlets, Tan Loc commune, An Loc commune, Loc Ha district;			
	 Hong Tan primary and secondary school (far from the construction site 100 m); 			
	- Tan Loc, An Loc primary school, Tan Loc Kindergarten			
	- Lu temple (far from the construction site about 30 m)			
	- Loc Ha General Hospital (located road side of Road No.22/12)			
	- My Loc Church – 100m from construction.			
SOCIAL ENVIRONMENT CON	DITIONS			
UXO	The proposed canal will be based on the existing route and during management and exploitation process so the potential for UXO is low. However, Ha Tinh PPMU has signed contract with No.319 UXO Clearance One Member Limited Company under Military Corporation			

⁷ The monitoring has been implemented in 25 May 2016 by Irrigation Science and Technical Implementation Institute

DATA ITEM	SUBPROJEC	Γ DATA			
	No.319 for UX	O clearance;			
Land use	Agriculture land is 1920 ha, and other land is 1729 ha ⁸ .Land is mainly used for agriculture development. Agriculture: mainly plant rice 2 crops/year;				
Nearest residential land	roads including and An Loc co	Residential areas are located along the canal and material transportation roads including Kim Tan and Tan Thuong hamlets, Tan Loc commune and An Loc commune. The distance between nearest residential houses and the proposed canal ranges from 20m to 200m;			
Rural infrastructure	lines running a existing secon hospital, medi These infrastr	Rural infrastructure works in subproject including the low-voltage electric lines running along Road No.22/12 road and other commune's road, the existing secondary and primary schools; kindergartens; Loc Ha general hospital, medical centres, public houses at communes/districts, etc. These infrastructures could be impact from construction as well as material transportation activities.			
Access to water supply & sanitation and solid waste management	Garbage is collected once a week in Tan Loc commune and Thinh commune. Garbage is transported to Loc Ha landfill located in a distance of 5 km from the construction sites.				
	There is 60% of people who access to sanitation latrines. Domestic wastewater discharges on the ground or to ponds, lakes and low positions. Households in the subproject area use storm water for domestic purposes.				
Agriculture	Agriculture: ma	ainly wet rice, b	ean, corn, veg	jetable;	
Population		lation density	are varied 4	27-1057 pers	29,458 people ons/km ² ⁹ . The
	An Loc	Binh Loc	Hong Loc	Tan Loc	Thinh Loc
	3393	4765	8582	5909	6809
Ethnic minorities	There are no e	thnic minority	groups living ir	the subprojec	ct area.
Livelihoods	+ In 2013, the main employment of the Loc Ha District is agriculture and handicraft production, occupying 64.32% of the local population. Trade and business shared 23.7% and industrial and construction was 11.8%.				
	+ The average	income is VNI	D 24 million/pe	rson/year;	
	+ The poverty rate of Tan Loc, Binh Loc, Hong Loc, An Loc and Thinh Loc communes are: 14.5%; 14.2%; 15.2%; 14.8%; and 16.2% respectively (following the new poverty line made by the Government)				
Physical and cultural heritage	There is no cultural heritage or preservation area in the subproject region.				
Public health	Diseases whic	h often occur i	n the summer	are diarrhoea,	petechial fever
	Besides, there	are respiratory	/ diseases like	sore throat, si	nusitis.

.

⁸ Source: PPMU of Ha Tinh province, SIP of lining the irrigation and drainage canal systems and road of the flooded area in Loc Ha district subproject, 2014

⁹ Source: PPMU of Ha Tinh province, SIR lining the irrigation and drainage canal systems and road of the flooded area in Loc Ha district, Ha Tinh Province, 2014

4. ENVIRONMENTAL IMPACT SCREENING

Table 3. Environmental impact screening

		POTEI	NTIAL IMPAC	т			
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE		
Pre-Construction Stage Impacts							
Disturbance of unexploded mine and bomb (UXO)	Yes	Minor	Negative	Temporary	Description: UXO can be left in some areas that had been suffered from war. Mine detector in subproject area may create safety risk for local people and workers. However, to help in securing safety for people, Ha Tinh PPMU has signed contract with No.319 UXO Clearance One Member Limited Company under Military Corporation No.319 for UXO clearance Location: Along the Khe Hao dam, irrigation canal of Khe Hao and Dong Ho reservoirs, subproject drainage canal and road systems. Objects: Local people living and cultivate in the subproject area. Affected level: Minor due to this affect will be temporary and can be controlled by hiring mine detector team. Time of impact: Temporary		
Effects on households from loss of residential or agricultural land	No				The canal and road is upgraded on the existing route. Thus, no households or land will be effected by the project		
Construction S	tage lı	mpacts					
Dust, vegetation clearing, noise, water quality or other impacts from development of borrow areas for construction materials	Yes	Minor	Negative	Temporary	Description: Stone, sand will be purchased from mines and quarry as mentioned above. The owners of these mines take responsibilities for any environmental problems related to vegetation clearing or water quality and PPMU; contractors need to check the permission of all suppliers before purchasing. Filled soil will be taken from Tung Son borrow pit, Tung Loc commune and Hong Loc borrow pit located at a distance of 7 km from the subproject site. The exploitation of sand and stones, operation of borrow pits could create dust, noise, water quality pollution and loss of vegetation. Event most exploiting activities is belong to third parties (as suppliers), but checking the compliance of these suppliers need to be implemented to ensure minimize negative impacts at these locations.		

		POTE	NTIAL IMPAC	т	
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE
					Location: Surrounding the borrow pits in Hong Loc and Tung Loc communes Affected level: Small, dust and noise will not be seriously affected because (i) the borrow pits locate on the base of the hills, surrounding by separated Acacia plantation area and cemeteries (ii) volume of construction material is not large, about 8020 m³ of construction stone, 30653 m³ of filling soil and large part of it will be utilized from excavation soils Time of impact: 24 months
Erosion or sedimentation caused during clearing or earthworks	Yes	Minor	Negative	Temporary	Description: In the work of excavating and filling for the proposed main weir canals, roads, construct canal structures such as culverts, water inlets and outlets, if excavated soil is not collected then siltation will be occurred. It will be able to cause stuck in other canals that pass through the proposed canals, in Khe Hao reservoir and create filling situation that affects cultivation areas of residents. Location: The intersections of the proposed canals and existing irrigation canals, Khe Hao reservoir, downstream of Khe Hao weir. Cultivation areas along the proposed canals and roads of the subproject Affected objects: Local people in Kim Tan, Tan Thuong, Tan Trung hamlets - Tan Loc commune who cultivate in the areas near the proposed canal. Affected level: minor due to excavated volume soil is reused to fill up banks of proposed canals and the construction for Khe Hao weir and the canals. The excavation and levelling activities will be scheduled to be implemented mainly in dry season; Time of impact: 24 months
Impacts arising from temporary storage areas including dust, noise and water quality	Yes	Minor	Negative	Temporary	Description: Dust could be arising from sand or soil stockpile if they are not covered carefully especially in dry condition. Runoff water could bring material from the temporary storage sites into the surrounding fields when its rain, impact on the water quality. Loading and unloading construction material at the temporary storage site could create noise and affect nearby households. Load and unload construction material could also create noise and dust and affect nearby people. Location: Temporary stockpiles for stone and sand in the subproject area. Affected objects: Water bodies near the

		POTE	NTIAL IMPAC	т	
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE
					temporary stockpiles like water rice fields, subproject canals. Local people living near temporary stockpiles. Affected level: Small, even the subproject stretches in a large area, the constructions sites are small and separated thus the temporary stockpiles are small and mobile. On the other hands, materials are mainly handled manually and the impact is only in the construction time so the impact will not be significant. Time of impact: 24 months
Pollution of waterways, aquatic environments or underground water from wastes, chemicals, waste water or disturbance of contaminated soils	Yes	Minor	Negative	Temporary	Description: In the process of excavating, dredging the canals, weir, road and filling banks of the canal, filling the road and constructing on canal structures, oil and grease leakage will be able to cause water and soil pollution. Wastewater from construction sites and worker camps could be discharged into surrounding water bodies such as the water rice fields in Tan Loc, An Loc, Hong Loc communes and impact on the water quality, reduce the productivity of the field. On the other hand, if waste such as spoiled soil, sand will not be collected and controlled strictly then it will able to make water turbidity increased. Location: At the downstream of Khe Hao dam, the proposed canals and other canals that receiving irrigation water from Khe Hao and Dong Ho reservoirs, and construction sites in Kim Tan, Tan Thuong, Tan Trung hamlets – Tan Loc commune and Thinh Loc, Hong Loc, An Loc commune; Affected objects: Water quality in the canal system of Dong Ho and Khe Hao reservoirs. Local people in the subproject area Affected level: Small, the affected level is insignificant because (i) the number of construction machines is small and the volume of construction water, (iii) the construction is scattered over the whole area of the subproject, therefore the concentration of the uncontrolled waste, oil and grease leakage is not remarkable; (iii) excavation and leveling activities are requested to be implemented during dry seasons; Time of impact: 24 months
Changes of	No				Upgrading the canals will not affect operation

		POTEI	NTIAL IMPAC	т	
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE
water quality due to changes in inlet operation which cause sedimentation or aluminiferous intrusion or pollution					of inlet as the main inlet from upper Khe Hao Reservoir is still the same, thus it will not cause sedimentation or aluminiferous intrusion or water pollution
Making sensitive flora disappeared and deteriorated	No				The proposed weir, canals and road are located on fields or near residential areas. The site clearance for construction of the canals only affects people's crops and weeds. There is no sensitive flora surrounding the subproject area.
Dust and air pollutant or exhaust fumes from construction equipment and transport activities	Yes	Minor	Negative	Temporary	Description: The construction activities will generate dust, air pollutants, particularly in dry/sunny days. Dust could also be generated from excavation activities and operation of trucks (both transporting materials and discarded soil, solid waste to Loc Ha landfill), and operation of mobile concrete mixing machines and other construction equipment. Location: Along the material transportation route (Tan Loc, Thinh Loc, An Loc, Binh Loc commune's roads and road no.22-12). Construction sites at Kim Tan, Tan Thuong, Tan Trung – Tan Loc commune and Thinh Loc and An Loc communes; Affected objects: Local people living near the construction sites and material transportation road. Affected level: Minor, due to i) very small and low capacity construction equipment and transportation truck will be mobilized on the site such as bulldozer 110Cv, excavator 0.4-0.8 m³, compactor 9 tons bulldozer 75CV; ii) the quality of machine must been registered, controlled and maintained periodically, iii) the construction site will be set up at a distance from the residential area. Time of impact: 24 months
Noise and vibration from construction equipment and transport activites	Yes	Minor	Negative	Temporary	<u>Description:</u> Construction activities such as clearance, excavation of foundation; transport of redundant spoils to the disposal site through unpaved roads cause the generation of noise and vibrations to be felt with the construction sites and adjacent areas. Noise

		POTEI	NTIAL IMPAC	т	
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE
					and vibration from construction equipment and material transportation truck when this is in operation will cause disturbance to workers and local people, especially the sensitive receivers such as school, kindergarten, hospital, and temple Location: Along the transport roads (the commune's road and road no.22-12) and construction sites at Kim Tan, Tan Thuong, Tan Trung – Tan Loc commune and Thinh Loc and An Loc communes; Affected objects: - The subproject workers - Local people in the subproject area. - Pupils at the Hong Tan primary and secondary school, Tan Loc primary school, Tan Loc kindergarten, An Loc primary school. Affected level: Affected level is minor due to (i) The number of construction machine is not many (03 bulldozers; 06 excavators; 06 tip trucks, 26 mobile concrete mixing machines) and distributes in large subproject area; (ii) construction sites are mainly on the field, away from residential areas. Time of impact: 24 months.
Increasing time and area of flooding	Yes	Minor	Negative	Temporary	Description: The subproject will upgrade 3374.5 m drainage canals in total. These canals also function as flood discharge canal for the subproject area. Close drainage water for canal construction could reduce capacity of the water flow discharge to Nghen river thus increases time and area of flooding in the flooding season. Location: Subproject area, especially the areas surrounding Dinh Lu – Cau Ngao and Loi Ma – Cau Coc drainage system in Thinh Loc and An Loc communes. Affected objects: - Cultivation activities in the subproject area - Local people who cultivated in the subproject area Affected level: The affected level is insignificant because of the short construction time (24 months) and the main construction activities for drainage canal upgrade will be implemented in dry season The water level of these drainage canals in the dry season is low as observed during the field trip. And the canal

		POTEI	NTIAL IMPAC	т	
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE
					totally could be closed section by section for construction. On the other hand, these two canals are only small drainage canal system and do not contribute the large part for floodwater discharge in the flood season. Time of impact: 8 months in the rainy season of 2 years construction duration).
Impact on traffic or conditions for property access	Yes	Minor	Negative	Temporary	Description: Material transportation will affect on movement demand, transporting goods of local people, disturbance to individual households and cause risk for safety traffic especially near the schools, kindergarten, Loc Ha Hospital, My Loc church Construction activities can affect on travelling for local people as increasing the density of traffic or block traffic movement. Temporary material storage at roadside, construction machines and construction waste could also block the movement of people who travel on the road. Upgrading weir, spillway, and canals could affect local people to access their cultivation area. Upgrading the management roads and transportation of materials will affect on movement demand, transporting goods of local people, disturbance to individual households and cause risk for safety traffic. Location: A long the material transportation roads including communal roads, road no. 22-12 and the proposed roads; cultivation field in the subproject area. Affected objects: Local people living in Kim Tan, Tan Thuong, Tan Trung − Tan Loc commune and Thinh Loc and An Loc communes. Student of Hong Tan primary and secondary school; An Loc primary school, patient of Loc Ha General hospital. Road users on the subproject roads (Loi Ma − Thien Thinh and Tan Loc − Binh Loc roads, Lower Khe Hao management road); and transportation roads (Tan Loc, Thinh Loc, An Loc, Binh Loc commune's roads and road no.22-12) Affected level: Small, since impact on traffic is minor because the traffic in the subproject area in low density with main transport vehicles of local peoples are motorbike and bicycle; people can use branch route in the communes to travel during construction. Time of impact: 24 months

		POTEI	NTIAL IMPAC	Т	
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE
Impact on public infrastructure such as communication electricity wires, intercommune road, etc.	Yes	Minor	Negative	Temporary	Description: Electricity cables were arranged in parallel with communal road to residential areas of An Loc, Thinh Loc, Tan Loc communes. Material transportation roads like Road No.22/12 and commune's roads themselves may be affected by the transportation activities. The existing canals of could also be affected by the construction activities. Location: Along commune's roads, road no.22-12, the proposed canals system of Dong Ho and Lower Khe Hao Reservoirs. Affected objects: Local people in the subproject area Affected level: Minor due to (i) small number of construction and transportation machines (03 bulldozers; 06 excavators; 06 tip trucks, 26 mobile concrete mixing machines); (ii) provincial road No.9, road No.22/12 and the cement concrete communal roads are in good condition; (iii) The impact is only happen in the construction phase so the impact is not large. Time of impact: 24 months
Employment or livelihood benefits from employment of local people	Yes	Minor	Positive	Temporary	Description: As agreed in the Loan agreement, contractors will use local labourers for simple works such as moving soils, give priority to poor families, female householders, woman if they need jobs. It aims to raise their income, create more jobs and contribute to poverty reduction for community. Location: The whole subproject area Affected objects: Local people in the subproject areas such as Tan Loc, Thinh Loc and An Loc communes. Affected level: Minor, this is a positive impact; however, it requires the coordination between the contractor and CPC of subproject communes and nearby communes in recruiting local labours (contractors often prefer to engage their own trained workforces rather than training unskilled labourers) Time of positive impact: 24 months
Effects on nearby heritage items such as graves, pagodas etc.	No				The subproject does not affect any national or local heritage items such as pagodas, temples nearby the proposed canals.
Construction	Yes	Medium	Negative	Temporary	Description: It is estimated about 300 workers

		POTE	NTIAL IMPAC	т	
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE
workers cause social disruption					will be mobilized on the construction sites. Workers have to get temporary residence certificate to avoid social disruption in the subproject area. Some social problems can appear such as gambling, drug addiction, prostitute, violence, conflict amongst workers, or between workers with local people. Location: Workers' camps along the proposed canal the roads and residential area at Kim Tan, Tan Thuong, Tan Trung – Tan Loc commune and Thinh Loc and An Loc communes; Affected objects: Local people in the subproject area. Affected level: Although the subproject is requested to hire local people and worker activities on the site can be controlled by working regulation in the construction site and construction duration is not long (within 24 months expected), this is still an average impact as the large number of workers (300 workers) and the contractor usually prefer to using their workers than the unskilled local people Time of impact: 24 months
Generation of excess spoil	Yes	Minor	Negative / Positive	Temporary	Description: Excess spoil could runoff into surrounding water and polluted surface water of the surrounding water bodies such as the water rice field in the subproject area. If temporary dumped at roadside, it could make the road dirty, slippery and create bad odour. The construction of the subproject will involve in excavate 23596 m³ soil. However, most of them will be reused for canal bank filling purpose. The small amount of excess spoil that could not be used for filling will be transferred to the dumping site of Loc Ha district. Local people usually take this spoil to fertilize their garden land. Representatives of the 5 commune CPCs could managed this activity to support local people fertilize their land. Location: Along construction sites in Kim Tan, Tan Thuong, Tan Trung – Tan Loc commune and Thinh Loc and An Loc communes; and the proposed weir site; Affected objects: - Local people who cultivate in the subproject area - Local people living near the construction areas - People who travel on the material

		POTEI	NTIAL IMPAC	т	
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE
					transportation roads - Pupils of Hong Tan primary and secondary schools; Tan Loc primary and kindergarten; An Loc primary school. Affected level: This is a minor impact because the estimated filling soil volume is 30653 m³ and most of the excavated soil will be utilized for filling. Only small amount of soil that the quality does not matched for filling purpose will be transferred to Loc Ha dumping site. Moreover, local people could take some part of the excess soil to fertilize their land. Time of impact: 24 months
Risk to health and safety to local people or construction workers	Yes	Medium	Negative	Temporary	Description: Dust, exhaust gas and noise generating from earthworks, transporting material, and operation of machines, etc, which could have direct affects on health of workers and local residents. Accidents may occur if during the construction, workers are not provided with safety equipment, obey construction regulations. Safety risks for local people could be since the illegal entering construction sites. Material transport may create the risk of affects on traffic safety and houses structure on road sides of provincial road No. 9, road No.22/12 and commune's roads in Kim Tan, Tan Thuong, Tan Trung – Tan Loc commune and Thinh Loc and An Loc communes; The temporary storage sites of material and waste, construction machines could block the movement, and increase the safety risks for local people and people who travel on the road, especially the sensitive receivers such as student of Hong Tan Primary and Secondary Schools; Tan Loc Primary and Kindergarten; An Loc Primary School, the patient of Loc Ha General Hospital, commune medical center, Lu temple and My Loc church. Concentration of workers with insanitation living condition may cause some respiratory, skins, eye diseases for local people as well as workers. Location: Residential areas located nearby the construction sites and material transportation roads (Provincial road No.9, Road No.22/12 and commune's roads, construction sites along the proposed canals. And workers those present on the site

		POTE	NTIAL IMPAC	т	
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE
					Affected level: This is a medium impact. The construction sites are mainly on the field, in open air with small quantity of construction volume, low transport frequency of trucks (8 trips/day) and short construction period (within 24 months as expected) so the dust and noise do not have remarkable affects on residents. However, the access road to the construction sites and the material transportation road (road No.22/12 and communal roads) go through residential area of several hamlets in Tan Loc, Thinh Loc, An Loc, Hong Loc communes with a lot of environmental sensitive points such as schools, kindergartens, hospital, medical centre, temple, church with higher safety risk during construction period. Time of impact: 24 months
Causes waste disposal problems from solid waste generated during construction activity or municipal waste generated in construction camps	Yes	Medium	Negative	Temporary	Descriptions: Solid waste that will be generated from construction mainly includes domestic waste and excavated soil. On estimation, about 1562 m³ of construction waste and 4500 kg domestic waste will be generated in a month during construction time. If they are not regularly collected and properly disposed, it could make odour and could be carried away by surface run-off into surrounding water bodies (mainly water rice cultivation field) and pollute surface water Location: Workers' camps along the proposed canals in Tan Loc, An Loc, Hong Loc Thinh Loc communes; cultivation area along subproject canals. Affected Objects: Construction workers and local people in the subproject area. Affected level: Medium, number of worker is 300 workers; construction activities include upgrade weir, spillway, irrigation and drainage canals and will generate about 1562 m³/month (construction waste) and 4500 kg/ month (domestic waste). There is a waste collected and transferred to Loc Ha dumping site — about 5 km from the subproject area. Domestic waste from workers will be also transferred to Loc Ha dumping site. Time of impact: 24 months
Impacts on irrigation activities	Yes	Minor	Negative	Temporary	<u>Description:</u> Construction of the proposed weir and upgrading the primary canals requires dry construction area, meaning of stop water flow in the existing primary canals. There will

		POTEI	NTIAL IMPAC	Т	
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE
					be a conflict between water demand for agriculture and construction demand during cultivation period and construction time; Location: The proposed canals and downstream cultivation area including Tan Loc, Thinh Loc, An Loc communes; Affected Objects: Cultivation field in the subproject area. Local people who cultivated in the subproject area Affected level: Small, Irrigation schedule could be changed flexibly to construction time; the design consultant has calculated the irrigation level for 2 crops during the weirs and canals construction is about 16000m³/ha in one year (estimated construction time for the weir and canal is less than 8 months) and the canal closed for construction section by section will not have large impact on the cultivation activities in the subproject area.] Time of impact: 24 months
Environmental Recovery	Yes	Minor	Negative	Temporary	Description: if construction sites are not cleaned up upon the completion of the canals construction waste and domestic waste from construction camps will pollute local environment. If contractors do not implement site restoration properly in accordance with environmental regulations then environmental problems such as soil erosion, sedimentation and accident may occur. Location: construction sites, Tung Son borrow pit, Tung Loc commune and Hong Loc borrow pit, Loc Ha landfill Affected objects: - People living near the proposed canals, camp sites, waste disposal and borrow areas - Land and water near the proposed canals, waste disposal and borrow areas. Affected level: Small Construction waste and domestic waste will be collected every week so the remains of the waste at the construction sites will be small. Campsites are assumed in small size and simple structure. Time of impact: temporary to permanent
Impacts in oper	ration	stage			
Inundation of	No				Construction of the weir and main canals, upgrading the primary canals and management

		POTENTIAL IMPACT			
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE
sensitive vegetation from operation of the canal					road will not cause flooding situation due to increment of irrigation area. On canal structures such as inlets, outlets and other controlling systems will be operated properly to handle flood. Moreover, there is no sensitive vegetation in the subproject area.
Excessive exploitation of surface water will make water supply capacity cannot catch up with demands and/or cause conflicts among households	No				Reasonably exploit the surface water according to approved design task of the works. Increase the ability to meet the water demand for agriculture, especially the area has not been watered for long time; thereby harmonize the conflicts between households using water.
Water quality is changed due to salinity intrusion, aluminiferous water or sedimentation	No				The subproject area is not affected by seawater or tide, so it is not affected by salinity intrusion or alumiferious; The proposed canals will be lined by concrete so the water will not be polluted by sediment.
Water is exploited at sensitive ecological places/or reservation areas	No				There is no sensitive ecological areas or protection areas in the subproject
Productivity is improved by increase of irrigation capacity	Yes	Significant	Positive	Permanent	Description: The completion of subproject will increase the irrigation area to designated capacity of Khe Hao and Dong Ho Reservoirs (from 64 to 228 ha for lower Khe Hao and 92 to 190 ha for Dong Ho Reservoir) Location: Tan Loc, Thinh Loc and An Loc, Hong Loc and Binh Loc communes. Affected Objects: cultivation area of the subproject; local people in the subproject area. Affected level: Significant
Cultivation habits will be changed due to the turning of land use for agriculture	Yes	Significant	Positive	Permanent	Description: The subproject will increase irrigation capacity of Dong Ho and Lower Khe Hao Reservoirs from current 49% and 28% respectively and the utilization area of agricultural land will be increased thanks to the sufficient irrigation water

		POTE	NTIAL IMPAC	Т	
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE
purposes					Location: The utilization area of agricultural land will be increased at less cultivated area where often occurs drought in Tan Loc, Thinh Loc, An Loc, Hong Loc, Binh Loc communes Affected Objects: cultivation area of the subproject; local people in the subproject area. Affected level: Significant
Reduce nutrition in the soil due to over-irrigation	Yes	Minor	Negative	Permanent	Description: There is not statistics or research in the area regarding the percentage loss of nutrients. Actually, the rate of soil nutrient loss is very small due to the cultivation in the plain with small slope that cannot cause drift of soil when it rains or excessive irrigation. After the proposed canals are completed, the regulating system will be facilitated and more flexible, hence, the land will not lose nutrients due to excessive irrigation. Location: beneficiary area in Tan Loc, Thinh Loc, An Loc, Hong Loc and Binh Loc communes Affected objects: - Soil quality of cultivated land - Local people in beneficiary area Affected level: Small
Soil erosion or scouring of streams or canals	No				The canal is built from concrete so that soil erosion and land slide will not occur;
Congested canals cause flooding situation	Yes	Minor	Negative	Permanent	Description: In case of improperly operation and regulation of the culvert system, water will cause overflows and broke the canals. In addition, waste on the canals can cause stuck in the canals. Location: Along the proposed canals. Affected objects: The subproject canals Local people in beneficiary area Affected level: Small

	POTENTIAL IMPACT					
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE	
Risks caused by natural calamity	Yes	Minor	Negative	Permanent	Description: If flooding water level and flow volume exceed the flood design frequency, the weir might be broken. Sediment at upstream of the weir makes water storage volume decrease and affects the weir structure; Location: at the weir and canals Affected objects: Weir and canal Affected level: Minor	
Affecting water quality due to the increased volume of pesticide or chemical used for water treatment or increasing the waste water	Yes	Minor	Negative	Permanent	Descriptions: After constructing the canals, the cultivated area will increase about 262 ha. Consequently, the quantity of pesticides or chemical fertilizers will be increased. The amount of pesticides on field surface and drainage system will affect the quality of and irrigation water, possibly groundwater. The risk will increase if the management of pesticides is not reasonable. Location: Irrigation area in Hong Loc, An Loc, Tan Loc, Binh Loc and Thinh Loc communes. Affected objects: Water quality in the subproject area Local people in the subproject area Affected level: Small due to application of advanced technology in agriculture and proposed training for local people such as IPM	
Affects on employment or livelihood	Yes	Significant	Positive	Permanent	Description: Increase incomes and living standards for people in the subproject area by means of increasing the agricultural land area for production, cultivation productivity and diversification plants of seasonal crops. Location: The subproject area in Tan Loc, Binh Loc, An Loc, Hong Loc and Thinh Loc communes. Affected objects: Local people in the subproject area Affected level: Significant	
Impacts on ethnic groups	No				There are no ethnic households living in the subproject area	
Generating solid and liquid waste	Yes	Minor	Negative	Permanent	Location: The subproject area in Kim Tan, Tan Trung, Tan Thuong – Tan Loc commune and Thinh Loc, An Loc, Binh Loc communes; Affected level: Small Agricultural waste after harvested or garbage from other farming activities such as pesticide covers and straw is popular. Currently, waste	

		POTENTIAL IMPACT				
IMPACT	YES/ NO?	MINOR OR SIGNIFICANT ?	POSITIVE OR NEGATIVE?	TEMPORARY OR PERMANENT?	BRIEF DESCRIPTION OF IMPACT LOCATION AND SCALE	
					has not polluted severely to the canal water quality, soil, crops and farming areas. However, in the long-term, it will affect increasingly to soil and water quality and plants if the management authorities do not control and minimize;	
Changing the service approaching ability of local residents thanks to building management roads for the work	Yes	Significant	Positive	Permanent	Management roads combined with traffic roads will increase the approaching ability of local residents to services from markets, areas for commodity and agriculture product exchange.	
Occupying land of canal bank and canal and weir safety corridor for individual purposes	Yes	Minor	Negative	Permanent	Location: the right of way (ROW) of the corridors of canals and dam of the sub-project in Kim Tan, Tan Trung, Tan Thuong – Tan Loc commune and Thinh Loc, An Loc communes. Affected level: Small The occupying land for planting perennial trees or for personal purpose in the right of way (ROW) of the canals and dam corridors are remains. Occupying land for planting perennial trees are violate regulations of the Ministry of Agriculture and Rural Development on protection corridor canals and makes difficulties in implementing compensation and project execution during upgrading of the work;	

5. OUTLINE ENVIRONMENTAL MANAGEMENT PLAN (EMP)

5.1. Environmental Mitigation Plan

Table 4 - Environmental mitigation plan

POTENTIAL IMPACTS	MITIGATION MEASURES	RESPONSIBILITY	COST			
Pre-constructi	ion stage					
High risk of environmenta I degradation due to inappropriate construction materials management planning	As planned in designs documents, the main construction material will be purchased from licensed borrow pits, quarries and mines near the subproject area. In case that the above material source will be changed, an appropriate material management plan should included the following: Required materials, potential sources and estimated quantities available Material supply manners: preferring to purchase from existing material quarries. Agreement with the local authorities Check with environmental permission/certification of the quarries to ensure that owners have considered environmental impacts and mitigation measures. Environmental recovery plan Material transportation manner plans and schedules Program for delivery of quarry and borrow material	Design Consultant; Ha Tinh PPMU	Included in the contract with design consultant No marginal cost			
High risk of environmenta I degradation due to inappropriate spoil and waste disposal plans	 Re-use of waste materials & spoil disposal locations included in bid and contract documents. Surplus soil including cutting humus is used for levelling the canal safety corridor; Select a properly treatment manners, preferred of for fill up the site of other projects activities/ purposes. Recheck Loc Ha dumping site before construction start. The expectation is that construction waste will be stored temporarily along the proposed road, domestic waste will be stored in rubbish bins and then will be collected and transfer to Loc Ha dumping site. Waste materials transportation manner plans and schedules Establishment of complaints management system for duration of the works 	Design Consultant; Ha Tinh PPMU	Included in the contract of design consultant No marginal cost			
Disturbance of unexploded mine and bomb (UXO	 Supervise and monitor the contract with Company No.319. Install signs and alarms system at the UXO removal area while implement to warn people from entering the area. Checking the construction site and UXO clearance certificate upon the UXO removal complete. 	Ha Tinh PPMU	Included in contract between Ha Tinh PPMU and UXO removal company			
Construction stage						
Dust,	■ Operation license of Tung Son and Hong Loc		No marginal			

POTENTIAL IMPACTS	MITIGATION MEASURES	RESPONSIBILITY	COST
vegetation clearing, noise, water quality or other impacts from development of borrow areas for construction materials	borrow pits have been approved by Ha Tinh DONRE. The operations licenses and approved environmental certificate need to be checked before starting purchase soil from these pits. In the exploitation process of Tung Son and Hong Loc borrow pits, Tung Loc and Hong Loc communes, contractor should follow environmental protection issues, including: Working machines must be under periodically quality controlled; Oil and other chemical pollutants from working machines should be strictly controlled and stored separately, avoiding leakages; Workers should use protective equipment while working within the Site; Temporary earth drainage system and ditch should be formed to store waste water safely in rainy season to reduce turbidity before releasing water into cultivated area; Water should be regularly sprayed within borrow areas to reduce dust generation Cover trucks that transport materials purchased from the borrow pits and mines (along provincial road No. 9, road No.22/12 and communal roads) Construction vehicles, machines and equipment need to meet standards of exhaust, noise, and vibration as regulated by the Government. The Contractor needs to submit the Engineer documents proving that all construction vehicles, equipment, and machines are checked and meet requirements concerning noise and vibration generation of the current Vietnam standards.	Contractor	cost and Included in the civil work contract
Erosion or sedimentation caused during clearing or earthworks	 Install sediment fences and/or sediment traps at intersections of the proposed drainage canal and the existing main drainage canals (intersections between Loi Ma - Cau Coc canal and Dong Doi main drainage canal; between Tan Loc - Binh Loc canal an Hong Tan main drainage canal) Construct temporary drainage canals to reducing affects on residential area in Tan Loc, Binh Loc, An Loc and Hong Loc communes; Upgrading main weir of reservoir need to be implemented section by section and provide vegetable cover as soon as possible when excavation completed. Avoid excavation activities during the rainy season where possible. 	Contractor	No marginal cost, and Included in the civil work contract
Impacts arising from temporary storage areas including dust, noise and water quality	 Store material location need to be at least 300 m away from the sensitive receivers such as Hong Tan primary and secondary schools, Tan Loc primary school and kindergarten, An Loc primary school, Lu temple Building up appropriate plan of material management to reduce material stagnated on the site; 	Contractor	No marginal cost and Included in the civil work contract

POTENTIAL IMPACTS	MITIGATION MEASURES	RESPONSIBILITY	соѕт
IIII AOTO	 Store material at high place with impervious surface and at least 10 m away from the water bodies (water rice fields, ponds, existing canals). Cover material stockpile with canvas when they are not in used and during rainy; Material transportation is strictly forbidden in the evening when stores are located next to residential areas. 		
Pollution of waterways, aquatic environments or underground water from wastes, chemicals, waste water or disturbance of contaminated soils	 Store chemicals in secure area, with concrete floor and weather-proof roof and at least 10 m away from water rice fields, ponds and existing drainage canals (Dong Doi and Hong Tan main drainage canals; Ensure construction equipment and vehicles are maintained in good conditions to avoid leakage; Use mobile sanitary toilets following regulations of Health Ministry and washing facilities at construction camps (Circular No 27/2011/TT – BYT promulgating on National technical regulation on Hygienic conditions for Latrines/ QCVN 01: 2011/BYT on National technical regulation on Hygienic conditions for Latrines) Provide rubbish bins and containers at camping sites and construction sites. Signed contract with local environmental company to regularly transport waste to the Loc Ha dumping sites; Management of hazardous materials and waste need to be followed the requirements which stated in Decree 36/2015/TT-BTNMT; Hazardous materials spilling out unexpectedly to be cleaned, reported and monitored; The wastewater and solid waste will not be allowed to through on surrounding water bodies (water rice fields, ponds, Dong Doi and Hong Tan drainage canals) 	Contractor	No marginal cost and Included in the civil work contract
Noise, dust or exhaust from construction equipment and transport activities	 Spray water at least one time per day in the dry/hot day in transportation road section goes through the sensitive areas (5 schools and kindergarten, Loc Ha General Hospital) Cover all trucks carrying raw materials to and from the construction area and along the transport routes as provincial road No. 09, road no.22/12 and the communal roads of An Loc, Tan Loc, Binh Loc and Hong Loc communes; Mobile concrete batching equipment to be arranged a distance of 500m away from residential areas of An Loc, Tan Loc, Binh Loc, Hong Loc communes and only allow operating during daytime. Operation schedule of noise generated equipment/vehicles need to be avoid the rest time of local people, especially in the location close to residential areas in An Loc, Tan Loc, Binh Loc and Hong Loc communes and along the transport 	Contractor Canvas, washing facilities	No marginal cost and Included in the civil work contract

POTENTIAL IMPACTS	MITIGATION MEASURES	RESPONSIBILITY	COST
	route of provincial road No. 9, road No.22/12 and communal roads; Provide adequate protective equipment for workers who run noise generated machines Equipment needs to meet standards of exhaust, noise, and vibration as regulated by the Government and they need to be properly maintained.		
Increasing the flooding area and time	 Inform construction schedule/ scope in advance to Tan Loc, An Loc, Binh Loc, Hong Loc and Thinh Loc CPCs. Discuss with agricultural and irrigation staff of Loc Ha DPC to identify suitable construction time. Complete construction section by section before start a new one to avoid long construction time. The construction of drainage canal will be implemented in dry season only. Construction equipment shall not be allowed to take place on the irrigation and drainage canals 	Contractor	No marginal cost and Included in the civil work contract
Affects on traffic or conditions for property access	 Inform construction schedule and scope in advance to Tan Loc, Binh Loc, An Loc, Hong Loc and Thinh Loc CPCs. Avoid material transportation in the rush hours when student go to schools, people go to work and back home (6h30 7h30 am; 11h30-13h30; 16h30-17h30. Install signal lamps and sign panels at crossing points with road branches or close to residential areas. Limit the speed of means of transport on the route; In case unavoidable block the access to schools, kindergarten, medical clinic the contractors should construct a temporary access route. Install traffic signals at site to regulate and limited speed of vehicles or signals of work site, especially at the intersections where closed to cultivation area of local people in Binh Loc, An Loc, Tan Loc and Hong Loc communes; Provide temporary access for local people if any bock of road happened during construction; Reinstate as pre-project condition for any damage caused by construction activities. 	Contractor	No marginal cost and Included in the civil work contract
Impact on public infrastructure such as communicatio n, electricity wires, intercommune road, etc.	 Consulting the sub-project engineering staff to identify public infrastructure and avoid or minimize physical impacts on their infrastructures; Work with local authorities to obtain agreement in using any public facilities; Limit the speed and load of material transportation truck in rainy season (April to October) to avoid accidents or damage road; Contractors must repair local roads, electricity lines and other local infrastructure if they are damaged by the construction activities. 	Contractor	No marginal cost and Included in the civil work contract

POTENTIAL IMPACTS	MITIGATION MEASURES	RESPONSIBILITY	COST
	 Warning and set rule for workers to follow traffic regulations (limit the velocity of trucks especially when travel in the sensitive areas of schools, kindergartens, hospital, temple); Install warning signs and avoid crashes to electric poles and houses. 		
Generation of excess spoil	 Utilise excavated spoil for filling purpose, as much as possible, to minimize the volume of excess spoil (estimated volume of filling soil is 30653 m³ and excavated soil is about 42200 m³); Temporary spoil disposal site shall be located at least 50 m from water bodies, like water rice fields, ponds, Dong Doi main drainage canal, Hong Tan main drainage canal, and shall be protected from erosion by avoiding formation of steep slopes; Unused excavated soil need to be transported to and disposed at Loc Ha dumping site. 	Contractor	No marginal cost and Included in the civil work contract
Construction workers cause social disruption	 Consult with Tan Loc, Binh Loc, An Loc, Hong Loc and Thinh Loc CPCs to arrange accommodation for workers (to avoid any negative impacts on local people's activities) and register temporary residence card for them; Recruit local people for simple work such as soil moving, road compaction Consider house leasing in locality in comparison with site camps and consultation with competent staff in planning local housing for workers in the local community. Request workers to collect waste as regulation and ensure that their construction camps are maintained in clean and hygienic conditions; Distribute guidance leaflets to workers or invite workers to disseminate about the prevention of infectious diseases and diseases that may occur in the areas of the shack camp. Raise workers' awareness of environmental sanitation, infectious diseases as well as prevention of HIV/AIDS and sexually transmitted diseases and dissemination on social evils like drugs, gambling, prostitution, violence, stealing, etc. 	Contractor	No marginal cost and Included in the civil work contract
Risk to health and safety to local people or construction workers	 Inform Tan Loc, Binh Loc, An Loc, Hong Loc and Thinh Loc CPCs in advance on construction schedule and scope Consult representatives of Hong Tan Primary and Secondary Schools, Tan Loc Primary School and Kindergarten, An Loc Primary School, Loc Ha General Hospital, Lu Temple and My Loc church on construction schedule and scope, working hour before commencement. Collaborate with infrastructure staff of the five CPCs to install warning sign boards, road hump (if applicable) near the sensitive areas; Provide workers with full safety equipment and train them how to use properly. 	Contractor Safety equipment	No marginal cost and Included in the civil work contract

POTENTIAL IMPACTS	MITIGATION MEASURES	RESPONSIBILITY	COST
	 Install mobile toilets at worker camps and provide sufficient domestic water 		
Causes waste disposal problems from solid waste generated during construction activity or municipal waste generated in construction camps	 Provide rubbish bins (02 bins at the main camp; 01 bin for each camp along the canal sides and its management road and request workers to collect waste and not to leave litter into any water resources; Collect used oil and grease in a place with cover roof and impervious ground and at least 10 m far from the water body; Collect solid waste regularly and transport to Loc Ha dumping site – 5 km away from the subproject area; Management of construction waste, domestic waste and hazardous waste as per proposed on waste management items. 	Contractor	No marginal cost and Included in the civil work contract
Impacts on irrigation activities	 The Contractor need coordinate with irrigation authority (Loc Ha irrigation exploitation management enterprise), the 5 commune's irrigation staffs and relevant cultivation households during upgrading Lower Khe Hao reservoir and the irrigation canal systems to reach agreement on water supply time (when construction suspension), construction time (should be implemented at the time when irrigation activities are not done); Upgrading Lower Khe Hao reservoirs and the irrigation canal system of Khe Hao and Dong Ho reservoirs in dry season with application of construction time when irrigation activities are not 	Contractor Ha Tinh PPMU, Contractor	No marginal cost and Included in the civil work contract No marginal cost Included in the contract

POTENTIAL IMPACTS	MITIGATION MEASURES	RESPONSIBILITY	COST
Impact by impropriate environmental recovery responsibilitie s	 Remove all of the construction machines and construction tools out of the construction sites upon construction complete. Upon construction complete, perform industrial clean at that site and temporary acquired land area and before hand over the land area back to local authorities Compensate adequately for the temporary acquired land area that could not be recovered Plant tree to recover the vegetation coverage. Taking photos of the clearance site before clearance and after recovery process complete to ensure the vegetation coverage has been recovered adequately. 	Ha Tinh PPMU Contractor	No marginal cost and Included in the civil work contract
Operation stag	ge		
Reduce nutrition in the soil due to over-irrigation	Coordinate with agriculture authority/ Agricultural extension units to ensure that farmers are trained on proper irrigation method;	DARD of Ha Tinh province	Provincial budget
Congested canals cause flooding situation	 Operate regulation works canal properly, flexibly with real condition to avoid waste and water overflow at the proposed canals; Periodically dredge irrigation canal system after harvesting crop, heavy rain and flood; Periodically inspect and maintain canals and on canal structures. 	Northern Ha Tinh Irrigation One Member Limited Company	Provincial budget
Risks caused by natural calamity	 Ensure that subproject design will meet all safety standards on prevention of flooding, storms and other potential natural calamity; Irrigation works exploitation factory need repairs and the works need periodic maintenance to ensure the operation capacity. 	Northern Ha Tinh Irrigation One Member Limited Company	No marginal cost Provincial budget
Affecting water quality due to the increased quantity of fertilizer or pesticide or chemical substances or waste water	 Coordinate with agriculture authority to ensure that farmers are trained on irrigation method; Cover, bottles from pesticide, insecticide as well as other substance such as herbicide should be stored in safety tanks at cultivation area before transport to dumping area; Ensure weed and other floating waste are periodically cleaned along the canals; Coordinate with Agriculture Extension Centre to ensure that farmers are trained on Insect Prevention Method (IPM). 	Northern Ha Tinh Irrigation One Member Limited Company/ DARD	Provincial budget and other assistance funds
Generated solid and liquid waste	 Periodically collect waste in canals and field surface; Build waste collection system; Improve the awareness of local people of waste management and collection in the field, canals via training courses. 	Northern Ha Tinh Irrigation One Member Limited Company/ DARD	Provincial budget
Occupation of	 Implement strict management measures to protect 	Northern Ha Tinh	Provincial

POTENTIAL IMPACTS	MITIGATION MEASURES	RESPONSIBILITY	COST
canal and canal safety corridor for individual purposes	the canal banks and reservoir protection area from being reoccupied (for tree planting and other encroachment) by regular check and establishment of regulations on treatment of violations (if any)	Member Limited	budget

5.2. Environmental Monitoring Plan

Environmental effects monitoring

7. Environmental effects monitoring is carried out to examine impacts of project in relation to ambient environmental conditions.

Table 5 - Environmental effects monitoring plan

Mitigation Measure	Parameters	Location	Methods	Frequency	Responsibilit y	Cost		
Construction	stage							
Minimization of noise generation	Noise level	At nearest residential areas of the proposed canal and residential areas near material	Observation and community consultation using equipment	Weekly, monthly or when community' s feedback is raised	Construction Supervision Consultant (CSC)	Included in the contract with Supervision contractor		
	roads in Kim noise dB Tan, Tan Trung, (A), if that is Tan Thuong – the big noise levels commune and Thinh Loc, An Loc communes;	transportation roads in Kim Tan, Tan Trung, Tan Thuong – Tan Loc commune and Thinh Loc, An	roads in Kim Tan, Tan Trung, Tan Thuong — Tan Loc commune and Thinh Loc, An	roads in Kim name of Tan, Tan Trung, (And Tan Thuong — the Tan Loc of Commune and was thinh Loc, An commune tand tan		Every 3 months during construction duration or if there is feedback about high noise levels from the community	Construction Supervision Consultant (CSC)/ Environmental Officer at PPMU	Included in the contract with Supervision contractor
						Every 6 months during construction period or when community's feedback is raised	Monitoring consultant on environmental safeguard policies of LIC team PPMU and CMPU environmental officers	Included in separated contract with CPMU
Minimization of dust generation	Dust concentratio n	The same monitoring locations of Noise	Observatio n and community consultation using equipment	Weekly, monthly or when community' s feedback	Construction Supervision Consultant (CSC)	Included in the contract with Supervision contractor		

Mitigation Measure	Parameters	Location	Methods	Frequency	Responsibilit y	Cost
			to measure dust level (TSP mg/m³) if high dust level was observed	Every 3 months during construction duration or if there is feedback about high noise levels from the community	Construction Supervision Consultant (CSC)/ Environmental Officer at PPMU	Included in the contract with Supervision contractor
				Every 6 months during construction period or when community's feedback is raised	Monitoring consultant on environmental safeguard policies of LIC team PPMU and CMPU environmental officers	Included in separated contract with CPMU
		The intersections of the proposed canals and existing canals; The intersections of	Visual Observatio n; Public consultation	Weekly and after large rain evens	Construction Supervision Consultant (CSC)/ Environmental Officer at PPMU	Included in the contract with PPMU
Control of	Sedimentati	the proposed canal and drainage canals.		Based on requirement of water supply	Local people, Community monitoring committee Local irrigation	Province budget
surface water quality	lubricating oil and solid waste				staff (commune) Environmental Officer at PPMU	PPMU operation budget
				Once every 6 months during construction or in case of at any time or in case of complaints	Monitoring consultant on environmental safeguard policies/LIC PPMU and CMPU environmental	Included in separated contract with CPMU

Upgrading irrigation/ drainage and rural road system in flooding-prone area of Loc Ha district Integrated Rural Development in Central Provinces Project

Mitigation Measure	Parameters	Location	Methods	Frequency	Responsibilit y	Cost
Labour safety and community safety	Number, use of labour equipment;	In construction sites and on material transportation	Observatio n and community consultation	of residents When community' s feedback is raised	officers Local people, Community monitoring committee	Without marginal cost
	signal system Obey for traffic law of transportatio n mean of construction material	roads along residential areas Kim Tan, Tan Trung, Tan Thuong – Tan Loc commune and Thinh Loc, An Loc		Weekly/ Monthly or when community' s feedback is raised	Construction Supervision Consultant (CSC)/ Environmental Officer at PPMU	Included in the contract with PPMU
	ттацетта	communes.		Once every 6 months during construction or in case of at any time if necessary	Monitoring consultant on environmental safeguard policies/LIC	Included in separated contract with CPMU
Operation sta	age					
Surface water quality	BOD, DO, pH, TSS, Total coliform; fecal coliform, turbidity	Location: At 04 points: at the drainage canals in Kim Tan, Tan Thuong hamlets, Tan Loc commune; at Khe Hao reservoir, at canal connected to the inlet	Observation and community consultation Or sampling methods following Vietnamese standard when receiving feedback from communities	Twice a year in two first years of operation (1 time in rainy season and 1 time in dry season)	DARD, Northern Ha Tinh Irrigation One Member Limited Company;	Included in operation cost of Northern Ha Tinh Irrigation One Member Limited Company
Waste managemen t	Conditions on environment al sanitation within the subproject area; temporary waste storage yard	Throughout the subproject area	Observatio n and community consultation	Once every 6 months in first 2 years of operation	Northern Ha Tinh Irrigation One Member Limited Company	Budget provided following regulations at Decree No.115
Periodical canal	Level of canal	Along the proposed canal	Field survey,	Once every 6 months in	DARD/ Northern Ha	Local budget

Mitigation Measure	Parameters	Location	Methods	Frequency	Responsibilit y	Cost
maintenance	sedimentati on and conditions of sluices, equipment and works on the main canal		community consultation	first 2 years of operation	Tinh Irrigation One Member Limited Company	
Re- occupation of canal corridor	Occupation area, type of occupation (for planting trees or other purposes)	Along the proposed canal	Field survey, community consultation	Once every 6 months in first 2 years of operation	DARD/ Northern Ha Tinh Irrigation One Member Limited Company	Local budget

5.3. Environmental Compliance Monitoring

8. Environmental compliance monitoring is carried out to test compliance with operating procedures, technical standards and/or contractor specifications in the EMP.

Table 6. Environmental Compliance Monitoring

Mitigation Measure	Parameters	Location	Methods	Frequency	Responsibility	Cost
Construction	Stage					
Environmenta Ily responsible procurement	Inclusion in bid docs	All canal alignment	Checking documents	Bid preparation, before start of civil works	Ha Tinh PPMU	Included in the design contracts
Plan construction materials management	Meeting minutes and agreement with local authorities	All canal and proposed route alignment	Checking documents	Prior to start of site works and throughout construction phase	Ha Tinh PPMU	Included in the design contracts
Plan spoil and waste disposal	Meeting minutes and agreement with local authorities	All canal and proposed route alignment	Checking documents	Prior to start of site works and throughout construction phase	Ha Tinh PPMU	Included in the design contracts
UXO removal	UXO clearance process and	Through out the construction	Checking documents/ Observation	Before construction start	Ha Tinh PPMU	There are separate bidding

Mitigation Measure	Parameters	Location	Methods	Frequency	Responsibility	Cost
	certificate	sites				packages
Construction	Stage					
Erosion and sediment controls	Condition and capacity of controls,	Throughout construction sites	Observation Document review and public	Weekly, monthly, quarterly, and After heavy rain	Environmental Consultant in CSC team and PPMU safeguard staff Local Community Monitoring Boards	Included in CSC contracts PPMU operation budgets Local budget
controls g o mitigation measures i Table 4	mitigation measures in		consultation	Once every 6 months during construction or in case of at any time if necessary	Monitoring consultant on environmental safeguard policies of LIC and CPMU safeguard staff	Included in contract with CPMU; CPMU operation budget
Materials storage	Materials storage areas Throughout construction review public	Throughout D	Observation Document review and	Weekly, monthly, quarterly	Environmental Consultant in CSC team and PPMU safeguard staff Local Community Monitoring Boards	Included in CSC contracts PPMU operation budgets Local budget
		public consultation	Once every 6 months during construction or in case of at any time if necessary	Monitoring consultant on environmental safeguard policies of LIC and CPMU safeguard staff	Included in contract with CPMU; CPMU operation budget	
Construction equipment and vehicles	Noise and exhaust generation; covering of trucks; oil/fuel leaks Implementin g of mitigation	Throughout construction sites	Observation document review and public consultation, interview workers on	Weekly, monthly, quarterly	Environmental Consultant in CSC team and PPMU safeguard staff Local Community Monitoring Boards	Included in CSC contracts PPMU operation budgets Local budget
	measures in Table 4		the site	Once every 6 months during	Monitoring consultant on environmental	Included in contract with CPMU;

		,	
Unaradina irriaation/ drainaga	and rura	al road avotom in	flooding-prone area of Loc Ha district
opgrauling irrigation/ urainage	anu nura	ai roau system m	nooding-profile area of Loc Ha district
11 115 15 1			
Integrated Rural Developmen	t in Centr	rai Provinces Pro	iect

Mitigation Measure	Parameters	Location	Methods	Frequency	Responsibility	Cost
				construction or in case of at any time if necessary	safeguard policies of LIC and CPMU safeguard staff	CPMU operation budget
Construction camp	Cleanliness; waste disposal facilities; general conditions All site	All site camps	Observation , interview workers on the site and documents review	Weekly, monthly, quarterly	Environmental Consultant in CSC team and PPMU safeguard staff Local Community Monitoring Boards	Included in CSC contracts PPMU operation budgets Local budget
conditions	And Implement mitigation measures in Table 4			Once every 6 months during construction or in case of at any time if necessary	Monitoring consultant on environmental safeguard policies of LIC and CPMU safeguard staff	Included in contract with CPMU; CPMU operation budget
Construction workers cause social	authorities cons	Throughout document construction review public		Weekly, monthly, quarterly	Environmental Consultant in CSC team and PPMU safeguard staff Local Community Monitoring Boards	Included in CSC contracts PPMU operation budgets Local budget
disruption			nt n			Once every 6 months during construction or in case of at any time if necessary
Risks to health and safety of local people and construction	Check implementat ion of all items. Check compliance to Labour Code of Vietnam	Throughout construction sites	Checking documents, observation and community	Weekly, monthly, quarterly	Environmental Consultant in CSC team and PPMU safeguard staff Local Community Monitoring Boards	Included in CSC contracts PPMU operation budgets Local budget
workers	and other relevant regulations		consultation	Once every 6 months during	Monitoring consultant on environmental	Included in contract with

Mitigation Measure	Parameters	Location	Methods	Frequency	Responsibility	Cost	
	and implement mitigation measures in Table 4			construction or in case of at any time if necessary	safeguard policies of LIC and CPMU safeguard staff	CPMU; CPMU operation budget	
Water pollution	Checking equipment certificate, surface water quality and	Through out construction sites	Checking documents, observation and community consultation	Weekly, monthly, quarterly	Environmental Consultant in CSC team and PPMU safeguard staff Local Community Monitoring Boards	Included in CSC contracts PPMU operation budgets Local budget	
	implement mitigation measures in Table 4			Once every 6 months during construction or in case of at any time if necessary	Monitoring consultant on environmental safeguard policies of LIC and CPMU safeguard staff	Included in contract with CPMU; CPMU operation budget	
Flooding	Checking construction schedule and implement		construction schedule Along	Checking documents, observation and community consultation	Weekly, monthly, quarterly	Environmental Consultant in CSC team and PPMU safeguard staff Local Community Monitoring Boards	Included in CSC contracts PPMU operation budgets Local budget
	mitigation measures in Table 4	canals		Once every 6 months during construction or in case of at any time if necessary	Monitoring consultant on environmental safeguard policies of LIC and CPMU safeguard staff	Included in contract with CPMU; CPMU operation budget	
Property access	Rehabilitate the possibility of temporary and fixed	Affected assets: roads in commune and affected assets	Observation and community	Once during construction works and once after finishing construction	Construction Supervision Consultant Local Community Monitoring Boards	Included in the Contracts signed with Construction Supervision and Loan Implementatio	
	access	COL		Every six months	Environmental Specialist of LIC Team	n Consultant Team (LIC) Local budget	

Initial Environmental Examination (IEE) Upgrading irrigation/ drainage and rural road system in flooding-prone area of Loc Ha district Integrated Rural Development in Central Provinces Project Mitigation Measure Parameters Location Methods Frequency

Mitigation Measure	Parameters	Location	Methods	Frequency	Responsibility	Cost
Waste disposal	Environmen tal sanitation at construction site and temporary waste storage	Throughout construction site	Checking documents, observation and community	Weekly, monthly, quarterly	Environmental Consultant in CSC team and PPMU safeguard staff Local Community Monitoring Boards	Included in CSC contracts PPMU operation budgets Local budget
	area and implement mitigation measures in Table 4		consultation n	Once every 6 months during construction or in case of at any time if necessary	Monitoring consultant on environmental safeguard policies of LIC and CPMU safeguard staff	Included in contract with CPMU; CPMU operation budget
Public infrastructure	Constructio n schedule, machine registration, warning signs board, speed limit	Throughout construction site	Checking documents, observation, public	Weekly, monthly, quarterly	Environmental Consultant in CSC team and PPMU safeguard staff Local Community Monitoring Boards	Included in CSC contracts PPMU operation budgets Local budget
	and implement mitigation measures in Table 4	ement consultation ation sures in	consultation	Once every 6 months during construction or in case of at any time if necessary	Monitoring consultant on environmental safeguard policies of LIC and CPMU safeguard staff	Included in contract with CPMU; CPMU operation budget
Impacts on irrigation activities	Checking construction schedule and implement mitigation	Throughout construction site	Checking documents, observation public consultation and working	Weekly, monthly, quarterly	Environmental Consultant in CSC team and PPMU safeguard staff Local Community Monitoring Boards	Included in CSC contracts PPMU operation budgets Local budget
	measures in Table 4	easures in v	with local authorities	Once every 6 months during construction or in case of at any time if	Monitoring consultant on environmental safeguard policies of LIC and CPMU	Included in contract with CPMU; CPMU operation budget

Mitigation Measure	Parameters	Location	Methods	Frequency	Responsibility	Cost
				necessary	safeguard staff	
Environmenta I recovery	Vegetable recovery on the sites and Implement mitigation measures in Table 4	Throughout construction area	Observation , public consultation, interview workers on the site	Completed Construction	Environmental Consultant in CSC team and PPMU safeguard staff Local Community Monitoring Boards	Included in CSC contracts PPMU operation budgets Local budget
Operation Sta	ge					
Waste management	Site cleanliness	Throughout sub-project area	Observation	6 monthly for first 5 years of operation	Northern Ha Tinh Irrigation One Member Limited Company	Provincial budget
Using irrigation water	Using matter	Households near canals	Observation and community consultation	Once every 6 months in first 5 years of operation	PPMU/ Northern Ha Tinh Irrigation One Member Limited Company	Budget provided following regulations at Decree No.115
Soil erosion or land slide in canal	Conditions of the canal; level of sludge in water	At sections which have not be rehabilitated	Observation	Once every 6 months in first 2 years of operation	PPMU/ Northern Ha Tinh Irrigation One Member Limited Company	Budget provided following regulations at Decree No.115
Prevention of soil erosion and land slide in the canal	Conditions of canal bank	At some representativ e locations in subproject area	Observation and community consultation	Once every 6 months in first 5 years of operation	PPMU/ Northern Ha Tinh Irrigation One Member Limited Company	Budget provided following regulations at Decree No.115

5.4. EMP Implementation Arrangements

Table 7 - EMP Implementation

		Roles and				sponsil	oilities			
Organization	Subproject Preparation		Subproject Implementation		Subpro	ject Op	eration			
CPMU	Advice	to	PPMU	Safeguards	Suggest	to	PPMU	Advice	to	PPMU

	Ro	les and Responsibilities	
Organization	Subproject Preparation	Subproject Implementation	Subproject Operation
	Officer on IEE and IEE/EIAR preparation Review and provide "no-objection" on IEE or IEE/EIARs submitted by PPMUs	Safeguards Officer on EMP implementation during construction Monitor progress during construction Consolidate environmental reporting from PPMU	Safeguards Officer on EMP implementation during first 2 years of operation Monitor progress during first year of operation Consolidate PPMU environmental reporting
Ha Tinh PPC	Sign-off on environmental assessment documents prior to submission for approval Approval of any subprojects requiring EIAR that are not subject to MONRE approval	Project owner with ultimate responsibility for environmental performance of subproject during construction	Project owner with responsibility for operation stage environmental performance including implementation of EMP during operation
Ha Tinh DONRE	Provide advice and guidance on environmental issues as required during subproject preparation	Monitoring implementation of EMP through their own internal monitoring system	Monitoring implementation of EMP through their own internal monitoring system
Ha Tinh PPMU	Engage consultant and have overall responsibility for IEE or IEE/EIAR preparation and submission for approval Ensure staff are adequately trained in environmental issues	Responsibility for EMP implementation during preconstruction and construction Ensure that contract specifications and bud documents include environmental requirements Undertake inspections and monitoring of environmental issues during construction Coordinate environmental monitoring reporting to CPMU	environmental issues
Loc Ha DPC	Approval of subproject EPPs in accordance with GOV legislative requirements	Monitoring implementation of EMP through their own internal monitoring system	Monitoring implementation of EMP through their own internal monitoring system
Loc Ha Subproject Support Teams (SST)	Assist in IEE/EPP preparation as required Assist PPMU to review bidding	Day to day supervision of contractors' in district including compliance with	

	Roles and Responsibilities				
Organization	Subproject Preparation	Subproject Implementation	Subproject Operation		
	documents, contract documents, and tenders to ensure environmental issues are adequately addressed	environmental management requirements Undertake environmental monitoring and coordination of local community environmental monitoring activities	coordination of local community environmental monitoring activities for first year of operation		
Commune Supervision Boards (CSBs) and local community members ¹⁰	Involvement in consultation and participation activities to identify and develop subprojects Ability to comment on environmental assessment documentation upon disclosure	Involvement in environmental monitoring activities under the direction of SSTs	Involvement in environmental monitoring activities under the direction of SSTs		
Construction contractor	n/a	Prepare detailed Site EMP to meet the Subproject EMP general requirements Allocate adequate resources to meet the requirements and obligations of Site EMP	n/a		
Northern Ha Tinh Irrigation One Member Limited Company	N/a	N/a	Monitoring mitigation measure in the operation phase		
LIC Team on environmental safeguard policies	N/a	Implement spot check environmental monitoring at subproject area once every 6 months. Monitoring results will be included in the report, which will be sent to CPMU.	N/a		
Construction Supervision Consultant	N/a	Implement construction supervision at construction sites every day. Implement environmental monitoring at subproject area every week. Monitoring results will be	N/a		

¹⁰ CSBs have been established under Decree 80 Regulation for Participatory Investment Supervision. Article 8 of Decree 80 provides the community with opportunities to inspect compliance, monitor implementation and evaluate the results of investments in the commune, including environmental impacts.

	Ro	les and Responsibilities	
Organization	Subproject Preparation	Subproject Implementation	Subproject Operation
		included in the report, which will be sent to PPMU.	

5.5. Monitoring and Reporting System Table 8 -Monitoring and Reporting System

Project Submitted Responsibility Type of Report **Frequency** Phase **To Whom** Site **Environmental** Construction Monthly Construction Ha Tinh **PPMU** Supervision Performance Report indicating Consultant compliance with Site EMP and monitoring results **EMP** Compliance Report Ha Tinh PPMU **CPMU** Quarterly compliance indicating with subproject EMP and monitoring results **EMP** Compliance CPMU/LIC Report Bi-annually or ADB indicating compliance with twice during subproject EMP and monitoring construction results depending on construction duration Subproject **Environmental** At completion of CPMU/LIC ADB Report subproject indicating overall subproject environmental performance **EMP** and compliance Operation **EMP Compliance Report:** Every six months ADB and Northern Ha **Operation** indicating compliance Tinh Irrigation for first two years Town and One Member with subproject EMP of operation. On-District commitments during operation going frequency to Limited People's be determined Committee Company based on review after 2 years

5.6. EMP Budget

Table 9 - EMP Budget

Item	Marginal Costs for Pre- Construction	Marginal Costs for Construction	Marginal Costs for Operation	Marginal Costs Sub- Total	
Monitoring					
Ha Tinh PPMU's Internal monitoring	Included in management cost of PPMU	Included in the Contracts with the Contractor and CMC as well as in PPMU's management cost	Local and provincial budget	Included in contracts or other operation capital sources	
Community monitoring	Not available (n/a)	Local budget (as in Decision No.80/2005/QĐ-TTg)	Local budget (as in Decision 80/2005/QD- TTg)	Local budget	
Independent monitoring consultant on environmental safeguard policies	n/a	Included in a separate contract with CPMU	n/a		
Training on capacity enhancement on environmental monitoring capability	n/a	Included in operation budgets of PPMU and CPMU	Local budget	n/a	
Public disclosure	Defined in consultancy contract on IEE	Included in a separate contract with CPMU	n/a	Public disclosure	
TOTAL (Intensify the capability and public disclosure)					

6. PUBLIC CONSULTATION AND DISCLOSURE ACTIVITIES

6.1. Description of Activities to Date

Table 10 - Public consultation and public disclosure activities

CONSULTATION METHOD	DE:	TAILS OF ACTIVITIES
Correspondence and meetings	•	15 /09/ 2014
with local authorities (District and Commune PCs, Commune Fatherland Front, Women's	Dates of meetings (if requested)	25/09/2014
Union, Youth Union and others)	Minutes of meeting attached (Yes / No)	Yes
Public meetings	Date(s) held	25/09/2014

CONSULTATION METHOD	DE	TAILS OF ACTIVITIES
	Location(s) held	PC's meeting hall and cultural house of Thinh Loc and Tan Loc communes
	Invitees	Commune PCs, stakeholders, village heads, Young Communist League, Fatherland front, Farmer Association, Women Union of the communes.
	Methods of invitation	Radio announcement and letter, coordinate with Women Union to mobilize women's participation in meetings
	Agenda attached (Yes / No)	Yes
	Minutes of meeting attached (Yes / No)	Yes
	Number of participants	Total have 81 people
		Man: 43 people

Women: 38 people

minutes of consultation)

(the list of participants will be closed in the

6.2. Outcomes of Public Consultation to Date Table 11 - Results of public consultation

Description of Issue Raised	By Whom?	Required Follow-up Actions?
Affect water supply and agriculture production	Local people	Make plan for water supply properly. Require farmer to implement in accordance with harvest. Inform local people about construction schedule
Dust	Local people	It is desired that all trucks have to be covered by canvas
Traffic safety on commune's roads	Local people	The Contractors are supposed to slow down when transporting materials by the residential area along commune's roads. It is necessary to plant construction signposts and speed limit signs.
Road damage	Local people	All trucks must follow the limited load capacity of the local roads. Contractors must repair material transportation roads if they damage them.
Construction workers cause social disruption and sanitation problems	Local people	It is requested that the workers maintain sanitation and public order; Register temporary residence card for workers; Request workers to collect waste as regulation and ensure that their construction camps are maintained in clean and hygienic conditions.

Description of Issue Raised	By Whom?	Required Follow-up Actions?
Generated solid and liquid waste	Local people	Improve the awareness of local people of waste management and collection in the field, canals via training courses.

6.3 Additional public consultation for the preparation of Environmental Impact Assessment

9. Additional public consultation activities have been implemented from 26 – 28 June 2016 at the meeting rooms of Binh Loc, Thinh Loc, An Loc, Hong Loc and Tan Loc CPCs. The participants are representatives of CPCs, village heads, Young Communist League, Fatherland front, Farmer Association, Women Union and local people from 5 communes. The discussion focus on potential environmental impacts during the construction phases and recommendation from local people for mitigations measures. Public consultations have been implemented by Irrigation Science and Technical Implementation Institute.

6.3. Future Public Consultation Activities

Table 12 - Proposed community consultation activities

Activity	Participants	Expected Outcomes	Schedule	Cost Estimate
Kick-off meeting prior to construction commencement	PPMU, the Contractor, Construction Supervision Consultant, community representatives at project area	Publicize construction contents, schedule and plan for water supply	1 week prior to construction commencement	Be estimated in EMP budget
Periodical meetings	Contractor, Construction Supervision Consultant and representatives of local authority, organizations and community at project area	Periodically check mitigation activities and arising problems Propose treatment alternatives and reach agreement on implementation	from construction commencement	Included in contract signed with parties

7. CONCLUSION AND RECOMMENDATIONS

- 10. The subproject "Upgrading irrigation/ drainage and rural road system in flooding-prone area of Loc Ha district" will be implemented by Ha Tinh PPMU under IRDPCP in Loc Ha district, Ha Tinh province.
- 11. An environmental assessment of the project has been carried and the main negative potential environmental impacts of the sub-project during construction and operation stages include:
 - Dust, vegetation clearing, noise, water quality or other impacts from development of borrow areas for construction materials;
 - Erosion or sedimentation caused during clearing or earthworks;
 - Impacts arising from temporary storage areas including dust, noise and water quality;
 - Pollution of waterways, aquatic environments or underground water from wastes, chemicals, waste water or disturbance of contaminated soils;
 - Dust and air pollutant or exhaust fumes from construction equipment and transport activities;
 - Noise and vibration from construction equipment and transport activities;
 - Increasing time and area of flooding;
 - Impact on traffic or conditions for property access;
 - Impact on public infrastructure such as communication electricity wires, intercommune road, etc.;
 - Construction workers cause social disruption;
 - Generation of excess spoil;
 - Risk to health and safety to local people or construction workers;
 - Causes waste disposal problems from solid waste generated during construction activity or municipal waste generated in construction camps;
 - Impacts on irrigation activities;
 - Improper environmental recovery.
- 12. A range of mitigation and monitoring measures has been developed for the sub-project, which includes the following activities:

Mitigation measures:

- Inform construction schedule and scope in advance to Tan Loc, Binh Loc, An Loc, Hong Loc and Thinh Loc CPCs
- Construction vehicles, machines and equipment need to meet standards of exhaust, noise, and vibration as regulated by the Government. Cover truck with canvas during the transport of construction materials, watering the road surface in the dry/hot day, especially in the borrow pits and mines area, near the schools and kindergartens, Loc Ha General hospital. Besides, other measures may include installing wheel washing equipment's at construction sites and regularly maintaining vehicles and machines.
- The Contractors do not transport materials at rest hours (7 pm to 6 am in the next day) and to be supposed to slow down when transporting materials by the residential area and to plan construction signposts and speed limit signs

- Provide rubbish bins to store domestic waste at the construction site; request workers not to leave litter; provide containers to store construction waste at construction sites; install sediment fences and/or sediment traps to collect sediment before it enters waterways;
- The Contractor should coordinate with irrigation authority (Northern Ha Tinh Irrigation one Member Limited Company), commune's irrigation staff and cultivation households in water supply area of the subproject to reach agreement on water supply duration (construction suspension), construction time (should be implemented at the time when irrigation activities are not done)
- During exploitation process, management authority should disseminate and consult the local authority/water users to limit the excessive use of water; establish regulated procedures and detailed water supply plan; update information year by year to inform users, implement dissemination and training on scientific irrigation to the community for understanding and implementation,
- Avoid deteriorating soil/water quality by increasing quantity of fertilizer and pesticide
 Ha Tinh PPMU should coordinate with Agriculture Extension Centre to ensure that
 farmers are trained on Integrated Pesticide Management (IPM)
- PPMU should ensure that subproject design will meet all safety standards on prevention of flooding, storms and other potential natural calamity

Monitoring activities:

In order to ensure the compliance of measures to mitigate negative environmental impacts caused by the subproject, these monitoring activities must be carried out:

- The contractor must implement measures to mitigate environmental impacts in residential areas near the construction sites, along the material transportation roads and construction sites, worker's camps etc. Their implementation can be monitored by observing and measuring water quality, air quality and frequency of implementing these measures. Moreover, the contractor must arrange adequate resources to meet general requirements and compulsory regulations on EMP at the construction sites;
- During operation stage, O&M agency (Northern Ha Tinh Irrigation one Member Limited Company) have to periodically manage water quality according to recent Vietnamese Standards and National Technical Regulations;
- PPMU should intensify the contractor's compliance with environmental regulations on material storage, construction equipment, waste disposal, air quality, dust, noise and vibration to ensure safety for the community during construction stage and operation stage; coordinate with local authorities to formulate and implement EMP.

Conclusion:

- 13. Upgrading irrigation/ drainage and rural road system in flooding-prone area of Loc Ha district, Ha Tinh Province will upgrade weir, spill weir of lower Khe Hao Reservoirs, upgrade irrigation canals of Khe Hao and Dong Ho Reservoirs, drainage canal and rural roads, ensure irrigation water supply for 418 ha cultivation area of Khe Hao and Dong Ho Reservoirs and improve livelihoods and reduce poverty for local people in the subproject area. It's also support the movement of local people and transportation of goods and drainage in the flood season. It is expected to directly benefit 29,458 people in 05 communes namely Thinh Loc, Tan Loc, Binh Loc, Hong Loc and An Loc. Thus, the subproject will contribute to promote socio-economic development and modernize rural area;
- 14. Negative environment impacts caused by the subproject mainly generate during the construction stage. However, these impacts are temporary and they will end when the canals are put into operation. Upon completion, the upgraded road will help supply irrigation water to farmers. On the other hand, it will bring positive impacts to the environment and promote

economic development for the subproject area. Thus, based on the Initial Environmental Examination, the consultants and Ha Tinh PPMU would like to recommend as follows:

- (i) There will not be any significant impacts to the environment and no further environment assessment is necessary.
- (ii) The IEE of the subproject "Upgrading irrigation/ drainage and rural road system in flooding-prone of Loc Ha district, Ha Tinh Province" should be approved by authorities so that next steps can be implemented to ensure good progress and project benefits.

8. **ANNEXES**

- Photos of subproject area and locations of air quality monitoring
- Public consultation and meeting minutes
- Photos of implementation of public consultation
- Data source

Annex 1: Photos of subproject area and locations of air and water quality monitoring



Photo 1: The existing spillway of Lower Khe Hao Reservoir



Photo 2: Overview of Lower Khe Hao Reservoir



Photo 3: Water quality monitoring point at Khe Photo 4: Air quality monitoring point at the Hao reservoir



beginning of canal no.3



Photo 5: Air quality monitoring point in front of Photo 6: Air quality monitoring point in front of Lu temple, Tan Loc commune



Hong Tan secondary school, Tan Loc commune



Photo 7: A section in Loi Ma - Cau Ngao Photo 8: Water quality monitoring point in Tan drainage canal



Thuong hamlet, Tan Loc commune



Management Road



Photo 9: Current condition of the Khe Hao Photo 10: Water quality monitoring point in Kim Tan hamlet, Tan Loc commune

Annex 2: Public consultation and meeting minutes

Meeting minutes at public consultation and lists of attendance in the public consultation meeting at Tan Loc commune

CỘNG HÒA XÃ HỘI CHỦ NGHĨA VIỆT NAM ĐỘC IẬP - Tự do - Hạnh phúc Tâm LàC, ngày LS tháng Đ năm DỰ ÁN PHÁT TRIỂN NÔNG THÔN TÔNG HỢP CÁC TỈNH MIỆN TRUNG (Loan 2357-VIE) BIÊN BẢN LÀM VIỆC Hôm nay, ngày LS tháng Đ năm 2014, tại Tam LaC
Dự ÁN PHÁT TRIỂN NÔNG THÔN TÓNG HỢP CÁC TÍNH MIỆN TRUNG (Loan 2357-VIE) BIÊN BẢN LÀM VIỆC Hôm nay, ngày 25 tháng 03 năm 2014, tại Tan Loc
BIÊN BẢN LÀM VIỆC Hồm nay, ngày 25 tháng 00 năm 2014, tại Tan Loc
I. Đại diện nhóm tư vấn của dự án Phát triển nông thôn tổng hợp miền Trung: - Ông/Bà. V. Long Thank Huếg, Chức vụ Chuyển gựa đã tượ - Ông/Bà. V. Long Lân Chức vụ Chuyển gựa tà tinh củ - Ông/Bà. V. Long lân Chức vụ Chuyển gựa mã thếy II. Đại diện Ban QLDA tinh - Ông/Bà. Ng Thank làn Chức vụ Cân bố kỷ - Ông/Bà. Chức vụ - Ông/Bà. Ng Ngạc Thach Chức vụ Chu trah ru Bư P xã - Ông/Bà. Ng Duy Đạn Chức vụ Chu tạch ru Bư P xã - Ông/Bà. Trần Thi Tịnh Chức vụ Chu tạch ha phụ nữ xã Nội dung làm việc:
- Ông/Bà V và chuy thanh thừ dy Chức vụ Chuyển gựa đã tro - Ông/Bà V và Hoàng Lân Chức vụ Chuyển gựa tài đi đi di - Ông/Bà Ng thanh lành - Ông/Bà Ng thanh lành Chức vụ Cảm bộ kỷ - Ông/Bà Chức vụ - Ông/Bà Chức vụ - Ông/Bà Chức vụ - Ông/Bà Ng Ng ac Thach Chức vụ Chu trich ru Bư Đ xã - Ông/Bà Ng Duy Đạn Chức vụ Chu trich ru Bư Đ xã - Ông/Bà Ng Duy Đạn Chức vụ Chu tạch ru Bư Đ xã - Ông/Bà Trân Thi Tinh Chức vụ Chủ tạch ha phụ nữ xã Ng Trong Thảo Nội dung làm việc:
II. Đại diện Ban QLDA tinh - Ông/Bà. Ng Thank lann. Chức vụ Can bỏ kỷ - Ông/Bà. Chức vụ - Ông/Bà. Chức vụ III. Đại diện địa phương - Ông/Bà. Ng Ng ac Thach. Chức vụ Chu trich u Bư D xa - Ông/Bà. Ng Duy Đan. Chức vụ chu trich u Bư D xa - Ông/Bà. Ng Duy Đan. Chức vụ chu trich ha phụ nữ xa Ng Trong Thảo Ng Trong Thảo Chức vụ Chu tạch ha phụ nữ xa Chức vụ Chu tạch ha phụ nữ xa Chức vụ Chu tạch ha phụ nữ xa Ng Trong Thảo Chức vụ Chu tạch ha phụ nữ xa Chức vụ Chu tạch ha phụ nữ xa Ng Trong Thảo Chức vụ Chu tạch ha phụ nữ xa
II. Đại diện Ban QLDA tinh - Ông/Bà. Ng Thank lann. Chức vụ Can bỏ kỷ - Ông/Bà. Chức vụ - Ông/Bà. Chức vụ III. Đại diện địa phương - Ông/Bà. Ng Ng ac Thach. Chức vụ Chu trich u Bư D xa - Ông/Bà. Ng Duy Đan. Chức vụ chu trich u Bư D xa - Ông/Bà. Ng Duy Đan. Chức vụ chu trich ha phụ nữ xa Ng Trong Thảo Ng Trong Thảo Chức vụ Chu tạch ha phụ nữ xa Chức vụ Chu tạch ha phụ nữ xa Chức vụ Chu tạch ha phụ nữ xa Ng Trong Thảo Chức vụ Chu tạch ha phụ nữ xa Chức vụ Chu tạch ha phụ nữ xa Ng Trong Thảo Chức vụ Chu tạch ha phụ nữ xa
II. Đại diện Ban QLDA tinh - Ông/Bà Ng Thank lann, Chức vụ Can bộ kỷ - Ông/Bà Chức vụ - Ông/Bà Chức vụ III. Đại diện địa phương - Ông/Bà Ng Ng ac Thach Chức vụ Chu tron u Bọ Dxa - Ông/Bà Ng Duy Đan Chức vụ chu tron u Bọ Dxa - Ông/Bà Ng Duy Đan Chức vụ chu tron họi tranh - Ông/Bà Trân Thi Tinh Chức vụ Chu tron họi hai phụ nữ xa Ng Trong Thảo Nội dung làm việc:
II. Đại diện Ban QLDA tinh - Ông/Bà Ng Thank lann, Chức vụ Can bộ kỷ - Ông/Bà Chức vụ - Ông/Bà Chức vụ III. Đại diện địa phương - Ông/Bà Ng Ng ac Thach Chức vụ Chu tron u Bọ Dxa - Ông/Bà Ng Duy Đan Chức vụ chu tron u Bọ Dxa - Ông/Bà Ng Duy Đan Chức vụ chu tron họi tranh - Ông/Bà Trân Thi Tinh Chức vụ Chu tron họi hai phụ nữ xa Ng Trong Thảo Nội dung làm việc:
- Ông/Bà. Chức vụ - Ông/Bà. Chức vụ - Ông/Bà. N.g. Ng.ac. Thach Chức vụ Chu trich U. B.N.D. Xa - Ông/Bà. N.g. Duy Đạn Chức vụ can bộ địa chính - Ông/Bà. Trần Thi Tinh Chức vụ Chủ tạch hai phụ nữ xa Ng Trong Thảo Nội dung làm việc:
- Ông/Bà. Chức vụ Chu trong - Ông/Bà. Ng. Ng. Qc. Thach, Chức vụ Chu tron 12 BND Xã - Ông/Bà. Ng. Duy Đạn, Chức vụ chu tron 12 BND Xã - Ông/Bà. Trần Thi Tinh Chức vụ chu tron ha phụ nữ xã Ng Trong Thảo Nội dung làm việc:
III. Đại diện địa phương - Ông/Bà. N.g. Ng. ac. Thach, Chức vụ Chu trưch ri BND xã - Ông/Bà. N.g. Duy Đạn, Chức vụ cu hó tra thinh - Ông/Bà. Trần Thủ Tinh Chức vụ Chủ trích há phụ nữ xã Ng Trong Thảo Nội dung làm việc:
- Ông/Bà. N.g. Ng. ac. Thach Chức vụ Chu tron ri BND xã - Ông/Bà. N.g. Duy Đạn Chức vụ chu tron là bô địa chinh - Ông/Bà. T. rấn Thủ Tinh Chức vụ Chu tron ha phụ nữ xã Ng Trong Thảo chủ trú họi này dan
- Ông/Bà. Ng. Duy Đạn. Chức vụ can bố đĩa chinh - Ông/Bà. Trầm Thị Tinh. Chức vụ Chủ tạch ha phụ nữ xã Ng Trong Thủo Nội dung làm việc:
- Ông/Bà. Ng. Duy Đạn. Chức vụ can bố đàa chinh - Ông/Bà. Trầm Thị Tinh. Chức vụ Chủ tạch hai phụ nữ xã Ng Trong Thảo Nội dung làm việc:
· · · · · · · · · · · · · · · · · · ·
Dai other bas quan ly du an got three guy mo
there and twig hang muc and du an thank phan
Though gia tax tink est got their we chick sach i but to tink at end who tax tro cong while and of
Chayen gow ma treday tound bay ve can take to
me trief cua du an va bien phap goan thiêu tac day má trười tray quai doan trước thước thiều thi con và vàn hành
Chuyin goa the two trink bay than van ba co

Sau khi đọc lại biên bản, những người có mặt đồng ý về nội dung biên bản, không có ý kiến gì Đại diện Ban QLDA tỉnh Đại diện UBND xã CÁC DU ÁN ODA Nguyễn Văn Dương Đại diện tư vấn SAFgaglen Vu Hoang Lân

CỘNG HÒA XÃ HỘI CHỦ NGHĨA VIỆT NAM Độc lập – Tự do – Hạnh phúc

Tan. Loc. ..., ngày 25 .tháng 09 .năm 2014

DỰ ÁN PHÁT TRIỂN NÔNG THÔN TỔNG HỢP CÁC TỈNH MIỀN TRUNG - KHOẢN VAY BÓ SUNG

BIÊN BẢN HỌP THAM VẤN CỘNG ĐỒNG

Về các chính sách an toàn: Môi trường, Tái định cư, Giới và Dân tộc thiểu số

Tên tiểu dự án: Sưa chưa : Xã Tan Lac, huyện	nang cap he they till tien viry ngap ung huyer Loc tla tinh tla vira
I. Thành phần tham dự:	
- Ong/Ba V. Ray Thanh Hilly,	Chức vụ chuyển giá đão tão
- Ong/Ba. Doan Van Dink,	Chức vụ Chuyển gựa thiết của
- Ong/Bà. Vi tloang lan,	Chức vụ Chuyển goa mà thờy
- Ong/Bà. Ng wyen thank Lain	Chức vụ Cân bộ ky thuất ban QL DAtha
- Ong/Bà Ng Ngoc Thuch ,	Chức vụ Chu trêh LBW DX à
- Ong/Bà Trân Thi Tinh,	Chirc vy. Chie tod has place not
- Ong/Bà. Ng. Trong Thao	Chức vụ Chủ từ th hà mày dan
- Đại diện những hộ bị ảnh hưởng	người, trong đónữ, chiếm(%), Dân tộc
thiểu sốngười, chiếm%	
II. Nội dung	

2.1 Các nội dung phổ biến:

- Cung cấp các thông tin về dự án như địa điểm, quy mô, các thông số kỹ thuật cơ bản
- Chính sách an toàn của dự án bao gồm: Chính sách về giới và sự tham gia của cộng đồng; Kế hoạch hành động giới; Chính sách môi trường, Chính sách Tái đinh cư và kế hoạch phát triển người dân tộc thiểu số.

2.2 Tham vấn cộng đồng:

- Tham vấn các vấn đề giám sát và tham gia của cộng đồng trong các giai đoạn chuẩn bị, thực hiện, vận hành tiểu dự án, các vấn đề về giới và lồng ghép giới, nhóm dễ tổn thương, hộ bị ảnh hưởng nặng...
- Tham vấn các vấn đề về môi trường, tác động môi trường tiềm năng của dự án bao gồm tác động lên môi trường tự nhiên và xã hội của khu vực dự án và những biện pháp giảm thiểu các tác động tiêu cực;

1

- Tham vấn các vấn đề về tái định cư, các tác động dự kiến, quyền lợi của người bị ảnh hưởng, các biện pháp giảm thiểu tối đa nhằm có ít tác động nhất đến người bị ảnh hưởng.
- Tham vấn nhu cầu đào tạo của các hộ bị ảnh hưởng.

III. Ý kiến thảo luận

III.1.Các vấn đề về giới, tham gia cộng đồng

Van con hain tring try nam kird mi a) Le suat tham care lop.

tap humin se bird try give a mor nam give tham gia the humin

- Do ingap hut, mine bare, mor triby aim vict new till phu mi

man bear phus throa con care

- lain con hims tring man give oughigh more to xay ra some caico,

xied und give dins ainfe wat hing the phu ui va hi em

- the rephi dins quigen co tring into hom try car vin che Van

othy gras phay their mat hay tai dring and, was ate misse card che

wan dry, then that car how they may dan, the phu ui try car heat dry

van dry, then that car hour day, the phu ui try car heat dry

van dry, then that car hour day and the phu ui try car heat dry

van dry, then that car heat dry a

III. 2. Các vấn đề về môi trường

Trong quai storm this cay, cong nhan phai thu gour was gin gin ve sina ma trulof, gin gan an me trait to train lay lan being tat ches down atte philey Xe cho vat lies phan tura three tai tray they also quea to gay he has atily. He cho vat lies phan to gay he has atily. He cho vat lies phan che phin bat, train but Troy goin to an van hand, ngush dan tog ture duy tri tot will then gou rac train ke rac ra kenh muley. Who that phus then gon has phan han ba dog phulang nos guy tink. Nha than phus then phan phan han ba dog phulang the van to an nine trat tog, we sinh to Thou, they has the gien the cay.

III.3. Các vấn đề về tại định cư và dân tộc thiểu số họ ho ho, tại được củ; No công trình trêy hanh trêy là thị từ nêy
co bay clong and hiring they day daiva
song trong cou vive du an
thic bies di an can than the theo naing
IV. Kết luận
che bes than gia than vai tang nhat che turing thin hier di an và mong cong tring son ste très land che le dan cho rana, vier très haul
Si du La trèp ving nhưng như cấy có bảy chả người dây trong boat trọng cant tac
di an som Live tiens hand.

CỘNG HÒA XÃ HỘI CHỦ NGHĨA VIỆT NAM Độc lập - Tự do - Hạnh phúc

Tan Loc., ngày 25 tháng 09 năm 2014

DỰ ÁN PHÁT TRIỂN NÔNG THÔN TỔNG HỢP CÁC TỈNH MIỀN TRUNG -KHOẢN VAY BỔ SUNG

DANH SÁCH ĐẠI BIỀU THAM DỰ CUỘC HỌP (Tham vấn cộng đồng về chính sách an toàn: Môi trường, Tái định cư, Giới và Dân tộc thiểu số

Tên tiểu dự án: Sĩa chưa này cấp hệ thấy trười tru vuy ngấp unghuyê lệc tha Xã. Tân Lộc huyên Lộc Ha tinh tha Tinh

STT	Họ và tên	Giới tính	Địa chỉ	Ký tên
1	thân Chi Cinh	กเงื	Chôm Kim Gân	2/2
2	Rayon Chi Chuy	_nu ~	Thon Fan trung	Johns
3	Ngujen Chi Nam	nu	Thom Kim Tow	783
4	elgyin thi Wuling	hu	Thon can trung	26
5	han Thi Thibing	nu	thon Kim Can	About
6	Nguyan Thi chi	nu'	Chon Can Chaing	Mild
7	Nguyên Chi Eu	nu?	Thon Kim Can	Tu
8	Nguyên Chi Chu	nu	Then tan Chaing	Afan
9	Kguyin Chi Hoa	ny	Phon Kim Tan	Hoa
10	Wayin thi Wong	nil	thôn Kim Tan	dinh / by
11	Ngujen Chi dinh	nů	Thon tein thanh	Dinh
12	Nguyãn Chi Hoan	ni	Thon kim Pan	Hoan
1000	hân Chi Lieu	mi	Then kim tan	Lin
	Nguyen Thi Chu	nu	Chon Tan Thường	US.
	Nguyễn Chi Hoài	ny'	Thon kim tan	alg
16	Abyuyen duy Doan	nam	Thon Kine Fin	Join
17		nam	Thon Lan tung	Djel

STT	Họ và tên	Glới tính		Địa ch		Ký tên
18	Nguyễn Định Ly	Mam	Thon	Tan	trung	ly
19	Não Due thai	Nam	Thon	Kim	Tân	Merce
20	Le van Hay	Nam	Chon	kim	Tan	Leever D.
21	Ngyen dinh Thao	nam	Thon	Can	tung	Than
u	Nguyễn xuân Linh	Nam	Chon	Kim	Tan	Linh
23	Nguyãn Georg Giron	Norm	Chon	Kim	Cân	Thurty
29	uguyên trong Than	Nam	Choc	Kim	Tan	Tout
25	Nguyên Duy đơn	Nam	Thee	kim	Tan	200 H
26	ehu van Veim	Nam	Chon	Kim	Tan	US
27	phon dinh phung		Thon (Tân	Thank	Λ.
24	Ngyen phi Hing		Chon	Tan	tung	Wants
29	Ngo Due Wing		Chon	Tan	thung	Hluss
30	Squen Doan Le	Nam		Kin	Tan	10
31	phan van Linh	nava	Thon	Tan	6 hiding	1 /11/39
32	Phan than van	Norm		Can	hug	All
33	Nguyên ngọc Phach	Nam		Thach	châu	700
329	Cu Chi Luc	nu	Thon	Tan	Chuong	Luc
35	Nguyễn Thi nga	nii	Phôn	Can	trung	Nga
36	Phan Bd Dan	nam			Chilong	tay
37	Nguyễn thi Chuẩn	กเรี			Thanh	Rinh
39	Nguyên Chi Binh	nú nú			Can	1 D
40					Can	Jan
41	Nguyon Định Tiến	nam		Tan Pan No	. 0	Col
12	Phen that Son		- 1-	tion 18		Oank Son

	Họ và tên	Giới tính	Địa chỉ	Ký tên
	Đại diện cộng đồng		Đại diện U	BND xã
			TX OF THE PROPERTY OF THE PROP	
			Advance of the second	
	Đại diện Ban QLDA tỉ	nh	Đại diện	tư vấn
(li	BAH QUÁN LÝ		8Alo	cenglan
ino o	BAH QUÂN LÝ CÁC DƯ ÁN ĐĐA		VI 7	englan loang Lan
Pino	BAH QUÀN LÝ CÁC DỤ ẢN OĐA PHƠNG NHIỆP (THE		VI 7	tu vån anglan loang Lan
End of	BAH QUÁN LÝ	g	Vú t	loang Lan
and a	BAH QUÀN LÝ CÁC DỤ ẢN OĐA PHƠNG NHIỆP (THE	g	Vú 7	loang Lân
Line of the second seco	BAH QUÀN LÝ CÁC DỤ ẢN OĐA PHƠNG NHIỆP (THE	g	Vú t	loang Lân
and a second	BAH QUÀN LÝ CÁC DỤ ẢN OĐA PHƠNG NHIỆP (THE	g	VI 7	loang Lân

Meeting minutes at public consultation and lists of attendance in the public consultation meeting at Thinh Loc commune

CỘNG HÒA XÃ HỘI CHỦ NGHĨA VIỆT NAM
Độc lập – Tự do – Hạnh phúc Thịnh Là, ngày & tháng \$.năm 20 DỰ ÁN PHÁT TRIỀN NÔNG THÔN TỔNG HỢP CÁC TỈNH MIỀN TRUNG (Loan 2357-VIE)
BIÊN BẢN LÀM VIỆC
Hôm nay, ngày 2. tháng . g. năm 2014, tại LABN.D×achúng tôi gồm:
I. Đại diện nhóm tư vấn của dự án Phát triển nông thôn tổng hợp miền Trung:
- ông/Bà. LE Thi Mông Phương Chức vụ. Từ Vain Chi
- ông/Bà Prêng De Chiar Chức vụ Tư Vain Tớc
- Ong/Bà. Ng. Thi Thu Trang. Chức vụ Trì vàn MT.
II. Đại diện Ban QLDA tỉnh
- Ông/Bà. Tran Van Son, Chức vụ Ban OL DA Tinh
- Ông/Bà Way Tunq, Chức vụ Từ Nain - thiế Kế
- Ông/Bà Chức vụ
III. Đại diện địa phương
- Ong/Bà. Nguyên Công Trinh, Chức vụ phố Ct UBNO Xã
- Ong/Bà Train Hilly Thin, Chức vụ CT thai nông dân
- ông/Bà Vo thị Tuyện Chức vụ phố CT Hai Phụ nư
Nội dung làm việc:
1) Ction thick ve So bo Di an, se ation thuse him Tow so Thinh Lat. 2) Til van this ke trumb bay ve cae thong so
2) Tis van thin ke trinh bay ve can thony so
Vy thurt Se allor that hier tai the philong
3) the van Chimh Saeh an town to he pho bien
moi dung Christin Sath to an Team của Diệ am
4) thi mhan mhưng of kain và kien nghị của
4) Athinton muling of Kilin va Kilin nghi của lanh đạo đưy phương và các hà BÁH

Sau khi đọc lại biên bản, những người có mặt đồng ý về nội dung biên bản, không có ý kiến gì Đại diện Ban QLDA tỉnh Đại diện UBND xã CÁC DU ÁN ODA Maurém Công Trinh Nguyễn Văn Dương Đại diện tư vấn Dittong Dire Chien

CỘNG HÒA XÃ HỘI CHỦ NGHĨA VIỆT NAM Độc lập – Tự do – Hạnh phúc

thinh Low ngày 25 tháng ... 9 năm 2014

DỰ ÁN PHÁT TRIỂN NÔNG THÔN TỔNG HỢP CÁC TỈNH MIÈN TRUNG - KHOẢN VAY BỔ SUNG

BIÊN BẢN HỌP THAM VẤN CÔNG ĐỒNG

Về các chính sách an toàn: Môi trường, Tái định cư, Giới và Dân tộc thiểu số

Tên tiểu dự án: Sác Aula. Xã. T.W.M. Lac., huyện.	mang cap he thong tust lieu voing ngép ung Lèc fra tinh He Tinh Lèc Ha
I. Thành phần tham dự:	
- Ong/Ba. LE. Thi Mong Philong	Chức vụ. TV Cpid
- Ong/Ba. Dilding Dile Chila	Chức vụ Tử VÃ TĐC
- Ong/Ba. Ng. Thi Thutrang	Chức vụ Từ Varo Mỹ Trường
- Ong/Bà Ilan Voin Son,	Chức vụ Ban QLOA
- Ong/Ba LE Vom Tung,	Chức vụ từ vàn thiệ tế
- Ong/Ba. Nguyen Cong Trinh	Chức vụ Pho CT UBND Xã
- Ong/Bà Vo Thi Tuyer,	Chức vụ. Pho et Hal phụ nư
- Đại diện những hộ bị ảnh hưởng thiểu sốngười, chiếm%	người, trong đónữ, chiếm(%), Dân tộc
II. Nội dung	
2.1 Các nội dung phổ biến:	

- Cung cấp các thông tin về dự án như địa điểm, quy mô, các thông số kỹ thuật cơ bản
- Chính sách an toàn của dự án bao gồm: Chính sách về giới và sự tham gia của cộng đồng; Kế hoạch hành động giới; Chính sách môi trường, Chính sách Tái đinh cư và kế hoạch phát triển người dân tộc thiểu số.

2.2 Tham vấn cộng đồng:

- Tham vấn các vấn đề giám sát và tham gia của cộng đồng trong các giai đoạn chuẩn bị, thực hiện, vận hành tiểu dự án, các vấn đề về giới và lồng ghép giới, nhóm dễ tổn thương, hộ bị ảnh hưởng nặng...
- Tham vấn các vấn đề về môi trường, tác động môi trường tiềm năng của dự án bao gồm tác động lên môi trường tự nhiên và xã hội của khu vực dự án và những biện pháp giảm thiểu các tác động tiêu cực;

- Tham vấn các vấn đề về tái định cư, các tác động dự kiến, quyền lợi của người bị ảnh hưởng, các biện pháp giảm thiểu tối đa nhằm có ít tác động nhất đến người bị ảnh hưởng.
- Tham vấn nhu cầu đào tạo của các hộ bị ảnh hưởng.

· III. Ý kiến thảo luận

III.1.Các vấn đề về giới, tham gia cộng đồng
1) Ban GS CD Se tuss thomb lap va se than
pa gram sar vao are quá himb của PA
2) Ban Ct SCD yeu can stilde atou tow, mong
case many liez co the this hish gram day
tit hon.
3) Hay phu nư và Chinh quyển địa phương
mhat the se tuan the mholing chinh color
Me himh dang crion va até nghi nha Thau
R new si dung nhan cong tại ơtia philông
mā Khong phán bia ve qua tinh
III. 2. Các vấn đề về môi trường
1) Lanh due die phildre what the you
a) Lawh dan dia philong what this you can the care van ati ve
mun truling mhis mu, tieng on dac hier là
We vas ste an town give thong va phai co
bien but the chair tai noi the cong
2) trong quá trinh sản xuất phái thông tim vớ
that quantly cong de be was dan con chi
atong do vos le hoach san xuar
3) trong quá trình vào hamb, ean phải tuần
This theo eac chinb Sach come mhis qui trinh
Vận hanb.

III.3. Các vấn đề về tái định cư và dân tộc thiểu số	
1) Do việc thuếc biên pự an đư thuếc hiên	
tien con ui tri cui, elu you la nome cap, do do-	
Le Khoney and huring stein tai Sais ena cac ho	
tou tia philong	
2) Land stew stea philding coun Kir se thing tim	
to cac he tai xa that gian và kà hauch thi	
Công lu thể để người dân chủ trong bệ trư that gọi cần xuất và chủ đồng tại Thu com các son phẩm năi có trên tuyện thủ Công	
car lan phan non co trên tuyên thi công	
3) Nhường Vàis ctế Khaic nhất Thong quá trinh thì cong lam hu hong hoạc thu họ nhưng tài Sán Của các hệ thí mhai phải họ	
tai Cin Cue Cat he - this only that place	
The One each list BAH	
IV. Kết luận	
1) Lamb das dia Philong Va eacho BAH	
Mhat tu ve vie hue huen DA tai dua phusing. 2) Me lanh oten this phusing se van tong va	
2) N lanh oten dia philong se van dong va	
thong tim to cae he BAH, ver cae Chimb Sach	
an toais cua prê aro eté ngres dan nom duse va this	_
hitis -	
1 New this cong tranh that grown san xuar mham han	
chế thiết hai. Nếu lavin anh hương stin các tài	
San Khac phai otor mi cho cae ho BAH.	
3	

CỘNG HÒA XÃ HỘI CHỦ NGHĨA VIỆT NAM Độc lập - Tự do - Hạnh phúc

Dự ÁN PHÁT TRIỂN NÔNG THÔN TỔNG HỢP CÁC TÌNH MIỀN TRUNG – KHOẢN VAY BỔ SUNG

DANH SÁCH ĐẠI BIỀU THAM DỰ CUỘC HỌP (Tham vấn cộng đồng về chính sách an toàn: Môi trường, Tái định cư, Giới và Dân tộc thiểu số

Tên tiểu dự án: Sula Chula nường Coip hữ thong thể tiêu Vũng ngấp Lộc Hà Xã Thính Lốc huyện Lộc Hà tinh Hà Từnh

T Họ và tên	Giới Địa chỉ tính	Ký tên
Nguyễn Văn Chinh	Xom 1 - Hong phoney	Nã Van Chin
Nguyễn Van Hoan Nguyễn Van Han	WT 2 WA NAT	hoan
Nguyễn Van Theu	Xom 3 - Xon Diem Xom 1 Hong Phong	- L'Lan
Vo Hong thyen	nt_	#
Ng. that Anh	yen Diêm	And
Vo Hong Lown	View Arleno	Punt
18 Thi Trang	Quang Trung	Thany
Phans Thi Linh	Hong phong.	Link
Vo thi Hao	Chicago Thing	x Hino
No Thi History	Chang Trung	Milon
Dischig Thi Limb	Hong phong	thy
Ng. Van Chian	Hong Phong	Chien
Vo Hong Dieg	Quang Trung.	March

STT	Họ và tên	Glới tính	Địa chỉ	Ký tên
	Dredny Thi Hiday		Orang Trung	V Hung
	Le Doan Pue		nt g	v Dué
	Tran the Him	120020	M	Stin
	Vo the line	1	Hong phone	249
	Vorthy Linh	Na	MT	1x Hud
	là Doan Mhan		yen Dies	v Vhan
	Vo Hong Chuan		Yen Oras	× Chuas
	Up Hong Lap		yen Dien	So brige 1
	phan The Vain		<i>'</i>	Mo
	ng Hai Anh		,	Bus
	Train Thi Trught		V.	Ansix
	Ng The Ser	Nu	y Dien	Sen
	to Thi Sy	Nu	Hong phong	lk
	ng Thi Tain	Nu	V. Phong	town
	Tran Qua Danh		Quary Trung	Dors
	vo Hong Lags		yen Diem	be
	During Thank Khiem		H. phony	wein
	Le Dan Ough		Quang Trung -	Tom
	Ng cong Trinh Trajo Huji Thia		PET. UB NO Xa	
			PET. UB NO Xã Chủ tích khối nong do	men
	nguin Dinh Rin		yen tim	27300
	Vo Thi Tuyer		yeo Diem	

	irin	Dja chi	Ký tên
Đại diện Ban QLDA tỉnh		Mguyên D Đại diện tư	
BAN QUÁN LÝ CÁC DỤ ÁN ĐẠA CHỆ VẬN THIỆN NHO ĐẠ CÁC ĐỤ ÁN ĐẠA CHẾ CHỐ NHO ĐỊCH THỊ CHẾ CHỐ NHO ĐỊCH THỊ CHẾ CHẾ CHẾ THỊ CHẾ CHẾ CHẾ THỊ CHẾ CHẾ CHẾ THỊ CHẾ		2 hu	
Nguyễn Văn Dương		Duding 5	Duc Chi

Photos of public consultation meeting





Photo 11: Public consultation in Tan Loc Photo 12: Public consultation in Tan Loc commune

Annex 3: Data Source

- 1. PPMU of Ha Tinh province, Subproject Investment Report (SIR) of lining the irrigation and drainage canal systems and road of the flooded area in Loc Ha district, 2014.
- 2. PPMU of Ha Tinh province, Basic Design Explanation of lining the irrigation and drainage canal systems and road of the flooded area in Loc Ha district subproject, 2014.
- 3. Tan Loc People's Committee, Annual Report on Social Economy, December 2013.
- 4. Thinh Loc Commune People's Committee, Annual Report on Social Economy, December 2013.

Annex 4: Environmental monitoring forms

Environmental Compliance Monitoring Form for Construction Package

Part A: General Project Information

Sub	project Name:			
SIR	Code: Subproject Package #: Activity Sector:			
Prov	vince: Districts:			
Des	ign and Supervision Consultant Firm:			
Con	struction Company Name: Contract Date:			
Con	tract Amount: Contract Duration (days)			
Pers	son Responsible: Phone			
PPN	//U EMO: Phone			
	Don't D. Manitaring about list			
	Part B: Monitoring checklist			
Per	formance Indicator 1. Design and Preparations			
_				
	PPMU to complete 1-4 in conjunction with the subproject design consultant at the	e time	the pro	ject is tendered.
Date	e of Monitoring:			
		Yes	No	Remarks
1.	Have all UXO been cleared prior to commencement of construction?	100	110	rtomanto
2.	Does the subproject design meet applicable engineering safety and public health			
	standards?			
3.	Have the resettlement provisions been disclosed to the affected communities and compensation made to affected persons or households?			
4.	For the applicable subproject type:			
	Doods such culture into invinction, would are also investoration, does the			
а	. Roads, embankments, irrigation works and coastal protection: does the design provide cross drainage to prevent flooding?			
b	. Markets: does the design provide washing facilities and toilets in the market			
	area?			
<u>The</u>	construction Supervision consultant (CSC) to complete 5-10 with the PPMU and	constr	uction	contractor at the
time	of start-up. Date of Monitoring:			
5.	Has the contractor prepared a Site EMP?			
6.	Has the contractor posted a public notice regarding the nature, extent and cost of			
7.	the project? Are locations for mixing plants sufficiently distant from houses, schools and			
۲.	hospitals?			
8.	Are agreements in place with owners for temporary use of land for worker camps and construction yards?			
9.	Have spoil disposal sites been selected in consultation with local authorities?			
	Are official permits on record for quarry sites and borrow pits?			
	Score (1-10; 10 total)			(%)

Initial Environmental Examination (IEE)

Upgrading irrigation/ drainage and rural road system in flooding-prone area of Loc Ha district Integrated Rural Development in Central Provinces Project

Performance Indicator 2. Worker Provisions

The CSC to complete	11-16 in	conjunction	with t	the PPMU	and	construction	contractor	following	commencemen	t of
construction. Date of M	1onitoring	l :						-		

	Yes	No	<u>Remarks</u>
11. Were local authorities consulted in the planning for the location of construction worker housing?			
12. Are supervisors or other site personnel trained in basic first aid emergency response measures?			
13. Are first aid kits readily available to workers at the job site along with instructions for use?			
14. Has the contractor or Inspector from the Department of Health undertaken an awareness program for communicable diseases/HIV-AIDS?			
15. Has the contractor provided safety equipment (hard hats, ear plugs, dust masks, safety boots and glasses) to workers and training in use?			
16. Are construction camps equipped with adequate water supply, sanitary toilets, washing facilities and facilities for waste collection and storage?			
Score (11-16; 6 total	l)		(%)

Performance Indicator 3. Biodiversity

The CSC should complete 17-21 in conjunction with the PPMU and construction contractor following commencement of construction. Date of Monitoring:

	<u>Yes</u>	<u>No</u>	<u>Remarks</u>
17. Does the project avoid encroaching on natural forests or wetlands?			
18. Does the project avoid adverse effects on flow of natural streams and water quality?			
19. Are worker camps located outside of forested areas and has the contractor restricted access of workers to forests, fishing and hunting?			
20. Does the contractor obtain fill materials only from pre-existing quarries, or from borrow pits within the strict limits of the construction zone?			
21. For irrigation sector projects, are effects on agricultural biodiversity limited through use of integrated pest management?			
Score (17-21; 5 total)			(%)

Performance Indicator 4. Community Based Monitoring

The CSC to complete 22 and 23 in conjunction with the PPMU and construction contractor following commencement of construction. Date of Monitoring:_____

	Yes	No	Remarks
22. Has the contractor posted a public notice regarding complaints from the			
community?			
23. Has there been a public consultation regarding construction, environmental			
impact, and the community complaints system?			
Score (22-23; 2 total)			(%)

Outcome of Public Consultation:									
Date:	Locatior	າ:							
Topics	covered	in	presentation:						
Comments fr	om Attendees:								

Initial Environmental Examination (IEE)

Upgrading irrigation/ drainage and rural road system in flooding-prone area of Loc Ha district Integrated Rural Development in Central Provinces Project

Performance Indicator 5. Community Values and Safety

<u>Items 24 – 35 should be inspected quarterly.</u> Date of Monitoring:

	<u>Yes</u>	No	Remarks
24. Is temporary access provided to adjacent properties as needed?			
25. Is permanent access to adjacent properties reinstated on completion of a segment of work?			
26. Are construction hours adjusted around houses, hospitals and schools to minimize disturbance?			
27. Does the contractor limit the scope of construction in progress to minimize community impacts?			
28. Are physical impacts on public infrastructure and service disruption minimized?			
29. Are materials transported on approved haul routes?			
30. Are construction equipments maintained in good condition?			
31. Do vehicles operate within legal speed limits?			
32. Are material loads traveling on public routes covered?			
33. Is dust suppressed by watering exposed surfaces?			
34. Has the contractor installed signs and lighting in vicinity of works on public roads?			
35. Is access to the construction site restricted to the public?			
Score (24-35; 12 total)			(%)

Performance Indicator 6. Hydrology/Water Pollution

<u>Items 36 – 43 should be inspected quarterly.</u> Date of Monitoring:

	Yes	No	Remarks
36. Are construction camps maintained in a clean and hygienic condition?			
37. Are oil, fuel and chemicals stored in enclosed areas (dyked or covered)?			
38. Is discharge of wastewater into water bodies used for water supply avoided?			
39. Is clearing activity suspended during rains?			
40. Does the contractor prevent discharge of concrete trucks to waterways?			
41. Have existing drainage patterns been maintained during construction?			
42. Are areas of standing water in the construction area drained and backfilled?			
43. Are sediment controls installed upslope of waterways?			
Score (36-43; 8 total)			(%)

Performance Indicator 7. Project Completion

Items	44	_	50	should	be	inspected	prior	to	finalizing	the	construction	works.
Date of	Monito	ring:_										

	Yes	No	Remarks
44. Have drainage fixtures, curbs, road shoulders and ditch slopes been finished out to prevent hazard to the public during use?			
45. Are ground surfaces in the project area graded to prevent water from collecting?			
46. Have all construction debris, tree cuttings, excess dirt, rubble and scrap been removed from the construction zone?			
47. Have all pits been filled in and graded to drain, underground tanks (including septic tanks) removed and holes backfilled?			
48. Are all waste products removed from the construction site, equipment yards and worker camps, including oil waste, scrap materials and equipment, building materials and domestic waste?			
49. Have all points of access (drives, walks) and utilities (water supply, power, communications) to public and private property been restored to original condition?			
50. Have all complaints by the local community and individuals been resolved by the Contractor?			
Score (44-50; 7 total)			(%)

Performance Tracking

Performance Tracking consists of three sections:

- a. Performance Follow-up, where performance shortfalls noted in prior monitoring are listed and checked against current monitoring results.
- b. Community Complaints, where issues raised by the affected community are registered, tracked and outcomes recorded.
- c. Performance Indicator Results, where environmental performance against indicators are recorded.

Section 1: Performance Follow-up

Column 1	Column 2	Column 3	Column 4	Column 5
Performance		Was agency	Was problem	Was performance
variable (#) /		responsible	corrected before	indicator
Date Observed	Reason for negative rating	notified? / Date	next monitoring?	adjusted?

Section 2: Community Complaints

Column 1	Column 2	Column 3	Column 4	Column 5
Person		Was agency	Was problem	Was Person
Registering		responsible	corrected before	satisfied with
Complaint / Date	Summary of Complaint	notified? / Date	next monitoring?	Action?

Initial Environmental Examination (IEE)

Upgrading irrigation/ drainage and rural road system in flooding-prone area of Loc Ha district Integrated Rural Development in Central Provinces Project

Section 3: Performance Indicator Results

Project Name:							SIR	No.:		Package
Project Start Date:										
	Startup	Rev.	Q1	Q2	Q3	Q4	Average	Completion	Rev.	Final
Recording Date:										
Design and Preparations										
2. Worker Provisions										
3. Biodiversity										
Community Based Monitoring										
5. Community Values / Safety										
Hydrology/Water Pollution										
7. Project Completion										
Submittal Date:		Fo	r Cale	ndar C	uarter	:				
Inspector:										
(Signature)										

Annex 5: Environmental mitigation measure to include into bid documents

POTENTIAL IMPACTS	MITIGATION MEASURES
	 Operation license of Tung Son and Hong Loc borrow pits have been approved by Ha Tinh DONRE. The operations licenses and approved environmental certificate need to be checked before starting purchase soil from these pits.
	• In the exploitation process of Tung Son and Hong Loc borrow pits, Tung Loc and Hong Loc communes, contractor should follow environmental protection issues, including:
	 Working machines must be under periodically quality controlled;
Dust, vegetation clearing, noise,	 Oil and other chemical pollutants from working machines should be strictly controlled and stored separately, avoiding leakages;
water quality or	- Workers should use protective equipment while working within the Site;
other impacts from development of borrow areas for	 Temporary earth drainage system and ditch should be formed to store waste water safely in rainy season to reduce turbidity before releasing water into cultivated area;
construction materials	 Water should be regularly sprayed within borrow areas to reduce dust generation
	 Cover trucks that transport materials purchased from the borrow pits and mines (along provincial road No. 9, road No.22/12 and communal roads)
	Construction vehicles, machines and equipment need to meet standards of exhaust, noise, and vibration as regulated by the Government. The Contractor needs to submit the Engineer documents proving that all construction vehicles, equipment, and machines are checked and meet requirements concerning noise and vibration generation of the current Vietnam standards.
Erosion or sedimentation	 Install sediment fences and/or sediment traps at intersections of the proposed drainage canal and the existing main drainage canals (intersections between Loi Ma - Cau Coc canal and Dong Doi main drainage canal; between Tan Loc - Binh Loc canal an Hong Tan main drainage canal)
caused during clearing or earthworks	 Construct temporary drainage canals to reducing affects on residential area in Tan Loc, Binh Loc, An Loc and Hong Loc communes;
Caranyonio	 Upgrading main weir of reservoir need to be implemented section by section and provide vegetable cover as soon as possible when excavation completed.
	Avoid excavation activities during the rainy season where possible.
	 Store material location need to be at least 300 m away from the sensitive receivers such as Hong Tan primary and secondary schools, Tan Loc primary school and kindergarten, An Loc primary school, Lu temple
Impacts arising from temporary storage	 Building up appropriate plan of material management to reduce material stagnated on the site;
areas including dust, noise and	 Store material at high place with impervious surface and at least 10 m away from the water bodies (water rice fields, ponds, existing canals).
water quality	 Cover material stockpile with canvas when they are not in used and during rainy;
	 Material transportation is strictly forbidden in the evening when stores are located next to residential areas.
Pollution of waterways, aquatic environments or	 Store chemicals in secure area, with concrete floor and weather-proof roof and at least 10 m away from water rice fields, ponds and existing drainage canals (Dong Doi and Hong Tan main drainage canals;
underground water from wastes,	 Ensure construction equipment and vehicles are maintained in good conditions to avoid leakage;
chemicals, waste water or disturbance of contaminated	 Use mobile sanitary toilets following regulations of Health Ministry and washing facilities at construction camps (Circular No 27/2011/TT – BYT promulgating on National technical regulation on Hygienic conditions for

POTENTIAL IMPACTS	MITIGATION MEASURES
soils	Latrines/ QCVN 01: 2011/BYT on National technical regulation on Hygienic conditions for Latrines)
	 Provide rubbish bins and containers at camping sites and construction sites.
	 Signed contract with local environmental company to regularly transport waste to the Loc Ha dumping sites;
	 Management of hazardous materials and waste need to be followed the requirements which stated in Decree 36/2015/TT-BTNMT;
	 Hazardous materials spilling out unexpectedly to be cleaned, reported and monitored;
	 The wastewater and solid waste will not be allowed to through on surrounding water bodies (water rice fields, ponds, Dong Doi and Hong Tan drainage canals)
	 Spray water at least one time per day in the dry/hot day in transportation road section goes through the sensitive areas (5 schools and kindergarten, Loc Ha General Hospital)
	 Cover all trucks carrying raw materials to and from the construction area and along the transport routes as provincial road No. 09, road no.22/12 and the communal roads of An Loc, Tan Loc, Binh Loc and Hong Loc communes;
Noise, dust or exhaust from construction	 Mobile concrete batching equipment to be arranged a distance of 500m away from residential areas of An Loc, Tan Loc, Binh Loc, Hong Loc communes and only allow operating during daytime.
equipment and transport activities	 Operation schedule of noise generated equipment/vehicles need to be avoid the rest time of local people, especially in the location close to residential areas in An Loc, Tan Loc, Binh Loc and Hong Loc communes and along the transport route of provincial road No. 9, road No.22/12 and communal roads;
	 Provide adequate protective equipment for workers who run noise generated machines
	 Equipment needs to meet standards of exhaust, noise, and vibration as regulated by the Government and they need to be properly maintained.
	 Inform construction schedule/ scope in advance to Tan Loc, An Loc, Binh Loc, Hong Loc and Thinh Loc CPCs.
Increasing the	 Discuss with agricultural and irrigation staff of Loc Ha DPC to identify suitable construction time.
flooding area and time	 Complete construction section by section before start a new one to avoid long construction time.
	The construction of drainage canal will be implemented in dry season only.
	 Construction equipment shall not be allowed to take place on the irrigation and drainage canals
Affects on traffic or conditions for property access	 Inform construction schedule and scope in advance to Tan Loc, Binh Loc, An Loc, Hong Loc and Thinh Loc CPCs.
	 Avoid material transportation in the rush hours when student go to schools, people go to work and back home (6h30 7h30 am; 11h30-13h30; 16h30- 17h30.
	 Install signal lamps and sign panels at crossing points with road branches or close to residential areas.
	Limit the speed of means of transport on the route;
	In case unavoidable block the access to schools, kindergarten, medical clinic the contractors should construct a temporary access route.
	 Install traffic signals at site to regulate and limited speed of vehicles or signals of work site, especially at the intersections where closed to cultivation area of local people in Binh Loc, An Loc, Tan Loc and Hong Loc communes;
	 Provide temporary access for local people if any bock of road happened during construction;

POTENTIAL IMPACTS	MITIGATION MEASURES
	 Reinstate as pre-project condition for any damage caused by construction activities.
Impact on public infrastructure such	avoid or minimize physical impacts on their infrastructures;
as communication, electricity wires, inter-commune road, etc.	 Work with local authorities to obtain agreement in using any public facilities; Limit the speed and load of material transportation truck in rainy season (April to October) to avoid accidents or damage road;
road, cto.	 Contractors must repair local roads, electricity lines and other local infrastructure if they are damaged by the construction activities.
	 Warning and set rule for workers to follow traffic regulations (limit the velocity of trucks especially when travel in the sensitive areas of schools, kindergartens, hospital, temple);
	Install warning signs and avoid crashes to electric poles and houses. Install warning signs and avoid crashes to electric poles and houses.
	 Utilise excavated spoil for filling purpose, as much as possible, to minimize the volume of excess spoil (estimated volume of filling soil is 30653 m³ and excavated soil is about 42200 m³);
Generation of excess spoil	 Temporary spoil disposal site shall be located at least 50 m from water bodies, like water rice fields, ponds, Dong Doi main drainage canal, Hong Tan main drainage canal, and shall be protected from erosion by avoiding formation of steep slopes;
	 Unused excavated soil need to be transported to and disposed at Loc Ha dumping site.
	 Consult with Tan Loc, Binh Loc, An Loc, Hong Loc and Thinh Loc CPCs to arrange accommodation for workers (to avoid any negative impacts on local people's activities) and register temporary residence card for them;
	■ Recruit local people for simple work such as soil moving, road compaction
Construction	Consider house leasing in locality in comparison with site camps and consultation with competent staff in planning local housing for workers in the local community.
workers cause social disruption	 Request workers to collect waste as regulation and ensure that their construction camps are maintained in clean and hygienic conditions;
	 Distribute guidance leaflets to workers or invite workers to disseminate about the prevention of infectious diseases and diseases that may occur in the areas of the shack camp.
	Raise workers' awareness of environmental sanitation, infectious diseases as well as prevention of HIV/AIDS and sexually transmitted diseases and dissemination on social evils like drugs, gambling, prostitution, violence, stealing, etc.
	 Inform Tan Loc, Binh Loc, An Loc, Hong Loc and Thinh Loc CPCs in advance on construction schedule and scope
Risk to health and safety to local people or	 Consult representatives of Hong Tan Primary and Secondary Schools, Tan Loc Primary School and Kindergarten, An Loc Primary School, Loc Ha General Hospital, Lu Temple and My Loc church on construction schedule and scope, working hour before commencement.
construction workers	 Collaborate with infrastructure staff of the five CPCs to install warning sign boards, road hump (if applicable) near the sensitive areas;
	Provide workers with full safety equipment and train them how to use properly.
	Install mobile toilets at worker camps and provide sufficient domestic water
Causes waste disposal problems from solid waste	 Provide rubbish bins (02 bins at the main camp; 01 bin for each camp along the canal sides and its management road and request workers to collect waste and not to leave litter into any water resources;
generated during construction activity	 Collect used oil and grease in a place with cover roof and impervious ground and at least 10 m far from the water body;

POTENTIAL IMPACTS	MITIGATION MEASURES
or municipal waste generated in	 Collect solid waste regularly and transport to Loc Ha dumping site – 5 km away from the subproject area;
construction camps	 Management of construction waste, domestic waste and hazardous waste as per proposed on waste management items.
	■ The Contractor need coordinate with irrigation authority (Loc Ha irrigation exploitation management enterprise), the 5 commune's irrigation staffs and relevant cultivation households during upgrading Lower Khe Hao reservoir and the irrigation canal systems to reach agreement on water supply time (when construction suspension), construction time (should be implemented at the time when irrigation activities are not done);
Impacts on irrigation activities	Upgrading Lower Khe Hao reservoirs and the irrigation canal system of Khe Hao and Dong Ho reservoirs in dry season with application of construction time when irrigation activities are not implemented. Construct Loi Ma - Thien Thinh, Tan Loc Binh Loc roads and Lower Khe Hao Management Road in the irrigation time instead.
	• If construction time could not avoid water supply time, divert canal must be built to transfer water for rice fields under the subproject area
	 Inform commune's irrigation staff, irrigation exploitation enterprise or relevant authorities in advance so they can make plan on their own initiative;
	 Ha Tinh PPMU and the Contractor need pay attention to mitigation measures to reduce damages or to implement compensation for arising impacts due to water supply cut off for construction activities.
	 Remove all of the construction machines and construction tools out of the construction sites upon construction complete.
Impact by impropriate environmental recovery	 Upon construction complete, perform industrial clean at that site and temporary acquired land area and before hand over the land area back to local authorities
	 Compensate adequately for the temporary acquired land area that could not be recovered
responsibilities	Plant tree to recover the vegetation coverage.
	 Taking photos of the clearance site before clearance and after recovery process complete to ensure the vegetation coverage has been recovered adequately.